

JGR Solid Earth

RESEARCH ARTICLE

10.1029/2021JB023851

Xin Zhong and Matthieu E. Galvez contributed equally.

Key Points:

- Inter dependence of element transfers in subduction zones
- Thermodynamics of intra-slab metasomatism of major elements
- Transport- and supply limited regimes of carbonate subduction

Supporting Information:

Supporting Information may be found in the online version of this article.

Correspondence to:

X. Zhong and M. E. Galvez, xinzhong0708@gmail.com; matthieu.galvez@gmail.com

Citation:

Zhong, X., & Galvez, M. E. (2022). The subducting slab as a chromatographic column: Regimes of sub-Solidus mass transport as a function of lithospheric hydration state, with special reference to the fate of carbonate. *Journal of Geophysical Research: Solid Earth*, 127, e2021JB023851. https://doi. org/10.1029/2021JB023851

Received 16 DEC 2021 Accepted 14 MAR 2022

© 2022. The Authors. This is an open access article under

the terms of the Creative Commons Attribution License, which permits use, distribution and reproduction in any medium, provided the original work is properly cited.

The Subducting Slab as a Chromatographic Column: Regimes of Sub-Solidus Mass Transport as a Function of Lithospheric Hydration State, With Special Reference to the Fate of Carbonate

Xin Zhong¹ (D) and Matthieu E. Galvez^{2,3,4}

¹Freie Universität Berlin, Institut für Geologische Wissenschaften, Berlin, Germany, ²Faculté des géosciences et de l'environnement, Institut des sciences de la Terre (ISTE), Lausanne, Switzerland, ³Institute of Geochemistry and Petrology, Swiss Federal Institute of Technology, Zurich, Switzerland, ⁴Geological Institute, Swiss Federal Institute of Technology, Zurich, Switzerland

Abstract We investigate the chemical budget of subduction zones at sub-solidus conditions using a thermodynamic-numerical simulation in which all major rock components are treated as soluble and potentially mobile in aqueous fluids. This new strategy significantly improves the accuracy of predicted fluid-rock equilibrium compositions in open petrological systems. We show that all slabs release volatiles and nonvolatiles to the mantle wedge, contributing to its refertilization. But some mobile constituents, such as alkali and alumina, may be trapped along layer boundaries or traverse without interaction depending on chemical contrasts between adjacent lithologies. The accumulation of igneous alumina and silica in the limestones of the centraleastern Pacific slabs drives their decarbonation and is marked by metasomatic garnet growth. Those slabs are also predicted to lose much of their alkalis before sub-arc depth. Even when they are produced in the altered mafic and ultramafic layers, fluids reach the slab/mantle wedge interface with distinct compositional signatures that are typical of the sedimentary cover. We distinguished supply and transport limited regimes of element subduction by testing the sensitivity of our mass balance to changes in slab hydration state (HS). Transport limited slabs sensitive to HS include notably a hotspot of carbon release to the mantle wedge (e.g., Costa Rica). Finally, we show that the quantitative budgets do depend on the geometry of fluid flows, and on assuming that slabs are mechanically continuous structures, which is questionable. Taken together, these insights will help better constrain the long-term chemical evolution of the shallow planetary interior, and the thermomechanical behavior of the subduction interface.

Plain Language Summary Subduction zones return chemical elements from the surface to the deep Earth. But quantifying this transfer has been challenging. Here, we present a model where all major elements are partly mobile in the fluid phase, enabling us to compile a chemical budget for subduction zones in which only fluids mediate mass transport. We identify transport and supply regimes of element subduction as a function of lithospheric hydration state, rock compositions, and slab temperature. We show that the transport of many rock-forming elements such as SiO₂, CaO, and Al₂O₃, within and out of the slab, modifies rock compositions has important implications to understand the role of slab-derived metasomatic fluid in modifying the composition of the mantle wedge over time, the mechanical properties of deeply subducted rocks, and it will inform future investigations for the high-pressure petrology of rocky planets in general.

1. Introduction

When oceanic lithospheres sink into the mantle in subduction zones, a fraction of their mass dissolves into mobile fluids that are released at increasingly higher pressure and temperature (*P-T*) conditions (Frezzotti & Ferrando, 2015; Hermann et al., 2013; Manning, 2004; Peacock, 1990; Schmidt & Poli, 1998; Stolper & Newman, 1994; Tatsumi et al., 1986). The chemical budget of a down-going plate and of the mantle wedge depends on the temporal and spatial sequence of this mass transfer (Kerrick & Connolly, 2001; Manning, 2004; Stolper & Newman, 1994). This applies particularly to the fate of carbonate (Ague & Nicolescu, 2014; Caciagli & Manning, 2003; Collins et al., 2015; Connolly, 2005; Gorman et al., 2006; Kerrick & Connolly, 2001a; Menzel et al., 2019; Sverjensky et al., 2014). But in systems composed of complex fluid and solid solutions, the fate of all



rocks' chemical components is often directly or indirectly connected (Brady, 1977; Ferry, 1979; Joesten, 1977). Hence, quantifying fluid-mediated mass transport in subduction zones has remained hindered by multiple experimental, theoretical, and computational challenges.

There is a broad range of numerical approaches to address the chemical budget of subduction zones. The first class is "geophysical" as it highlights the importance of mechanical forces and density contrasts in the transfer of mass at convergent plate margins (e.g., Gerya & Meilick, 2011; Hetényi et al., 2007; Malavieille & Ritz, 1989; Platt, 1986; Tackley, 2000). But the mechanical properties of rocks depend on their composition, in particular their volatile content. Therefore, another class of approaches, thermopetrological, have prioritized accurate descriptions of advective mass transport, and most quantitative subduction zone budgets of H₂O and CO₂ to date have been obtained from them (e.g., Connolly, 2005; Gorce et al., 2019; Gorman et al., 2006; Hacker, 2008; Kerrick & Connolly, 2001a; Konrad-Schmolke et al., 2016; Tian et al., 2019; Van Keken et al., 2011). In particular, the use of molecular CO₂-H₂O fluid activity-composition models calibrated using phase equilibria (Aranovich & Newton, 1999; Holland & Powell, 2003; Kerrick, 1974) have ensured reliable estimates of water and C partitioning within fluid-rock systems over a wide range of *P*-*T* conditions. But the thermodynamic consistency of those fluid models with dehydration and decarbonation equilibria does not guarantee their absolute accuracy in prediting fluid compositions. For example, they assume that silica and other nonvolatile rock components (e.g., CaO, MgO, and K₂O) are insoluble, or infinitely compatible with the rock. This, of course, is not the case in reality (Manning, 2004; Schneider & Eggler, 1986; Spandler et al., 2007). We later refer to this inaccuracy as the "infinite compatibility assumption." It is a primary reason why conventional thermopetrological tools have remained limited to account for fluid-driven high-P/T metasomatic processes (Ague & Nicolescu, 2014; Angiboust et al., 2014; Bebout & Barton, 1993; Galvez et al., 2013; Penniston-Dorland et al., 2012; Sorensen et al., 2010; Vitale Brovarone & Beyssac, 2014) and their role on the fate of carbonate (Galvez & Pubellier, 2019; Kelemen & Manning, 2015; Plank & Manning, 2019).

The fate of carbonate in subduction zones is tied to the silicate environment: silicates produce the water solvent when they dehydrate, and they also control the solubility of carbonates. This link is commonly illustrated by the simplified decarbonation equilibria such as (Aranovich & Newton, 1999; Harker & Tuttle, 1956; Kerrick, 1974):

$$SiO_2 + CaCO_3 = CaSiO_3 + CO_2$$
(1)

Although it is not shown, Equation 1 is catalyzed by water at typical sub-solidus subduction zone conditions (Aranovich & Newton, 1999; Kerrick & Connolly, 2001a). If the silica constituent is assumed to be only a part of the rock, the reaction may either proceed from *P*-*T* driven (i.e., prograde) changes in the chemical potentials (μ) of CaCO₃, CaSiO₃, and/or SiO₂ (Aranovich & Newton, 1999; Harker & Tuttle, 1956), or via fluid-driven modification of μ (CO₂) by metasomatism, which drives Equation 1 to the right or left (Ferry & Dipple, 1991; Rumble, 1982). But real subduction fluids contain dissolved solutes (Frezzotti & Ferrando, 2015; Manning, 1998; Philippot & Selverstone, 1991; Rustioni et al., 2021; Schmidt & Poli, 2003; Schneider & Eggler, 1986; Spandler et al., 2007) and those fluids are therefore neither strictly binary, nor purely molecular. Hence, changes in chemical potentials (μ) of the constituents in Equation 1 may also result from diffusion or advection of SiO₂, CaO, MgO and/or other major rock components in and out of the system (Abart, 1995; Ague, 2003; Ferry et al., 2011). This can be expressed by rewriting Equation 1 as:

$$SiO_2(aq) + CaCO_3 = CaSiO_3 + CO_2$$
⁽²⁾

where $SiO_2(aq)$ denotes that silica may also be part of a reactive fluid phase. Accounting quantitatively for this mutual dependence of element mobility has long remained elusive in the numerical modeling of subduction zones processes. This is in large part due to the lack of computational tools to connect the mobility of volatile and nonvolatiles at a more fundamental level.

In this context, Sverjensky et al. (2014) have called attention to the capacity of the HKF equations of state (Helgeson et al., 1981), as implemented within the Eq3/6 software package (Facq et al., 2014, 2016; Wolery, 1992), to deal with solvent-electrolyte solutions owing to the knowledge of water dielectric constant over an extended range of *P-T* conditions (Fernández et al., 1995, 1997; Pan et al., 2013). But extensions of electrolyte chemistry to high-*P/T* petrological systems must account for two of their most defining characteristics: the existence of mixed volatile fluids characterized by variable water activities, and the prevalence of multicomponent minerals and solid-solutions at those conditions. Galvez et al., (2015) introduced a simple approach to achieve that goal. It takes advantage of the precision of phase equilibria calculations by Gibbs energy minimization (e.g., Connolly, 1990) to compute the speciation and elemental stoechiometry of mixed-volatile high-*P* fluids using a limited set of chemical potentials, regardless of mineralogical complexity (Galvez et al., 2015, 2016). Their metric of relative acido-basicity Δ pH reflects the fundamentally hybrid character of high-*P*/*T* fluids, simultaneously electrolytic and molecular. But the chief advantage of this approach is that it relaxes the "infinite compatibility assumption" of conventional petrological models. The integration of this new method within the PerpleX software by Connolly and Galvez (2018) provided a flexible numerical plateform for applications related to lithospheric devolatilization/revolatilization (Menzel et al., 2020; Piccoli et al., 2019) or the nature of fluid-rock interactions on Enceladus (Daswani et al., 2021). Connolly and Galvez (2018), in particular, reported that the mobility of potassium in fluids released from the model GLOSS sediment (Plank & Langmuir, 1998) may lead to its early depotassification, and therefore dehydration. But this result may have been amplified by infiltrating the sediment with initially pure H₂O, which calls for a systematic approach of sub-solidus mass transfers where all lithologies, sediment, mafic, and ultramafic would be partly soluble in the fluid phase.

Building on these premises, we report the first attempt at a quantitative chemical budget of subduction zones using a thermopetrological numerical method in which all rock components have a finite and experimentally constrained solubility in the fluid phase. We pay special attention to the process of slab decarbonation. Because subduction zones differ in composition and thermal structure, we run each individually when relevant data are available. For the sake of simplicity, we set up a model which considers dissolution, fractionation, and vertical transport of aqueous fluids and their dissolved solutes within, and out of, subducting slabs. We test the sensitivity of our chemical budgets quantitatively by adopting a range of upper mantle hydration states. Our model relies on a set of geological, chemical, geodynamical, and computational assumptions that are discussed and tested quantitatively when possible. The concepts and methods discussed here are a step forward in understanding the role of fluid-rock interactions on the composition and mechanical behavior of the Earth's lithosphere, and by extension the chemical dynamics of other rocky planets as well.

2. Methods

2.1. Chemical Structure of the Subducting Slab: Model Setup

Our slabs are composed of four layers: a sedimentary layer underlain by a basaltic upper crust, a gabbroic lower crust, and a partially serpentinized upper mantle layers at the bottom of the pile (see an example in Figure 1). The composition of the sediment is from Plank (2013), including H_2O and CO_2 , with a few minor modifications to ensure consistent Ca/C ratio when C abundances are lacking, or particularly uncertain (Figure S1 in Supporting Information S1). The altered basalt (1.5 km thick) and gabbro (5.5 km thick) compositions are from Pearce (1976). It has been shown that seafloor alteration of the oceanic lithosphere enriches the top basaltic layer in potassium (Alt & Honnorez, 1984; Staudigel et al., 1989). To simulate this process, we considered a 300 m thin layer containing 1.5 wt% K₂O, underlain by 1.2 km of basalt with a concentration of 0.1 wt% (Supplementary Table S1 in Supporting Information S1). Overall, this corresponds to a K_2O content of the altered basalt layer of around 0.37 wt%, which is consistent with natural observations (Alt & Teagle, 2003). The altered upper mantle (AUM) composition is from Hart and Zindler (1986), and its thickness and water content is a variable and discussed below. The global input of CO_2 associated with the basalt + gabbro (igneous) layer is ca. 85.8 Mt CO_2 yr (23 Mt C/yr), and it is around 7.5 Mt CO₂/yr (2.1 Mt C/yr) for the altered upper mantle. The total CO₂ trench input is ca. 186 Mt CO₂/yr with 92.6 Mt CO₂/yr from the sedimentary layer, which reflects the global estimate of Kelemen and Manning (2015), that is, 147–242 Mt CO₂/yr. The Makran subduction zone is not included in our simulations because no thermal model was available (Syracuse et al., 2010). Results of a model with higher C contents in the slab (ca. 267 Mt CO₂/yr from Clift, 2017) are also presented for comparison in the supplementary materials.

Because H_2O is the main solvent in our model, we have adopted a range of hydration states for our slabs. Our trench sedimentary H_2O composition is from Plank (2013) and the mass of subducted sediment per year is from Clift (2017), leading to a trench sedimentary H_2O input ca. 269 Mt H_2O/yr (Table 1). However, about 74 Mt H_2O/yr is lost as a free-fluid phase at the first grid of our simulation. This reduces the effective sedimentary input flux to ca. 195 Mt H_2O/yr (Table 2). When combined with the igneous input of 638 Mt H_2O/yr (Jarrard, 2003), the sedimentary and igneous flux in our models is 833 Mt H_2O/yr (Table 2). This is consistent with the general





Figure 1. (a) Thermal structure of the Cascadia subduction zone (redrawn from Syracuse et al., 2010). Layers are composed of sediment, basalt, gabbro, and peridotite. The cartesian coordinate system is transformed into in-slab depth coordinate and slab surface depth coordinate for modeling purpose. (b) Schematic depiction of the modeling approach. The rock undergoes dissolution first during dehydration reaction and the volatile and dissolved nonvolatile components are fractionated from the bulk rock. The fluid is then transported to the grid above in the in-slab coordinate. (c) Match between magnetotelluric data (McGary et al., 2014) and H₂O flux based on 2D dissolution-fractionation-transport model for the Juan de Fuca plate below Cascadia. Three fluid pulses are highlighted. (d) Fluid flux (H₂O) as a function of the surface depth. The fluxes at the surface of the sediment and Moho are shown with different colors. The subduction length is 850 km for the Juan de Fuca plate beneath Cascadia (Clift, 2017). The integrated H₂O flux in (d) at the slab surface is ca. 28 Mt/yr down to 120 km.

consensus on the amount of H_2O structurally locked in sediment and altered igneous minerals (Hacker, 2008; Jarrard, 2003; Rüpke et al., 2004; Van Keken et al., 2011), that is, ~720 Mt H_2O /yr.

The main uncertainty is associated with the serpentinized AUM on a global scale. In detail, Schmidt and Poli (2003) propose an upper-estimate AUM input of ca. 860 Mt H₂O/yr, while the upper-estimate of Rüpke et al. (2004) is ca. 1,200 Mt H₂O/yr (see Table 3 in Van Keken et al., 2011). We opted for a conservative value, and our high hydration state (HHS) model assumes an AUM input flux of ca. 1,522 Mt H₂O/yr, or 2 wt% H₂O in 10 km of AUM (linear vertical variation of H₂O content from 0 to 4 wt%). This gives a total of 2,429 Mt H₂O/yr, which is similar to the global input flux of 3,000 Mt H₂O/yr calculated by Cai et al. (2018) based on regional seismic refraction studies in Mariana. Hacker (2008) estimates a lower AUM input of ca. 570 Mt H₂O/yr. Our intermediate hydration state (IHS) model matches this estimate, with ca. 608 Mt H₂O/yr in the AUM, which leads to a total trench input of ca. 1,516 Mt H₂O/yr (Tables 1 and 2), that is, an average of 2 wt% H₂O over a 4 km AUM layer using a linear variation scheme from 0 wt% (AUM bottom) to 4 wt% (AUM top). For our low-end hydration state model (LHS) reflects this consensus with an AUM input of 258 Mt H₂O/yr or 0.85 wt% of H₂O over a 4 km AUM layer using the same linear variation scheme. The total trench input in this case is ca. 1,166 Mt H₂O/yr (Table 1 and 2). Our global estimates may locally overestimate (Canales et al., 2017) or underestimate (Cai

Table 1

Global Water Budget Derived From Our Models and Implications for Sealevel Variations in the Phanerozoic

	H ₂ O HHS (Mt/yr)	H ₂ O IHS (Mt/yr)	H ₂ O LHS (Mt/yr)	Arc H ₂ O (Mt/yr)	Arc H ₂ O (Mt/yr)	Return flux max 360 m (Mt/yr) ^g	Return flux 100 m (Mt/yr) ^g	Return flux 0 m (Mt/yr) ^g
Reference	This model	This model	This model	PM12	F08	PM12	PM12	PM12
Sediment ^a	269	269	269					
Igneous ^b	638	638	638					
AUM	1522°	608 ^d	258°					
Total	2,429	1,516	1,166					
Lost <20 km ^f	74	74	74					
Lost 20-80 km	145	141	138					
Lost 80-150 km	1,015	730	561	324–990	650			
Return flux past 20 km	2,355	1,442	1,092					
Return flux past 80 km	2,210	1,301	954					
Return flux past 150 km	1,195	571	393			576-702 (630)	306-432 (360)	252
Sea-level change over 542 Myr (m)	870–996	246-372	68–194					

Note. The return fluxes (mantle input) are compared to different sea-level change scenario predicted by Parai and Mukhopadhyay (2012). PM12 refers to Parai and Mukhopadhyay (2012) and F08 to Fischer (2008). Numbers in parenthesis denote an average.

^aThis is calculated from sediment compositions of Plank (2013). ^bFrom Jarrard (2003). ^cFrom Schmidt and Poli (2003). ^dFrom Hacker (2008). ^cCorresponds to the low end from Rüpke et al. (2004); Van Keken et al. (2011) and low end of Schmidt and Poli (2003). ^fThis refers to the H₂O lost at the first grid during phase equilibrium computation. ^gThis indicates the return fluxes associated with 3 different sea-level decline scenario in the past 542 Myr based on Parai and Mukhopadhyay (2012), and 0 m indicate no sea-level change. The sea-level change is calculated based on our computed return flux and the return flux vs. sea level change relationship from Parai and Mukhopadhyay (2012).

et al., 2018) the water input for some hot or cold slabs, respectively. This point has been tested on a case by case basis when possible.

As a first-order approximation, pressure within the subducting slab is lithostatic using an average rock density of 3,300 kg/m³, and slab thermal structures are from the numerical results of Syracuse et al. (2010) using input data of plate convergence rate and slab geometry. Slab names, length, and sediment thicknesses are taken from Clift (2017). Most of our article refers to the extraction efficiency of a given oxide *i*, it is denoted by η_i . This variable corresponds to the fraction of the trench input of component *i* released from the slab within a certain depth interval. The subduction efficiency $\sigma = 1-\eta$, describes the fraction of the total input retained within the slab to a given depth (Galvez & Pubellier, 2019; Volk, 1989). The extraction and subduction efficiency of carbon is a key parameter of long-term climate and biogeochemical models (Caldeira, 1991; Hoareau et al., 2015; Volk, 1989). For our purpose, the forearc region starts at 20 km, the depth of the first grid of our model. It ends at 80 km, unless stated otherwise in our sensitivity tests. The "sub-arc region" includes the forearc and ends at a depth of 150 km.

2.2. Computational Strategy

Simulating the processes of internal thermodynamic equilibration between fluids and solids by Gibbs energy minimization requires that the system is closed, that is, its composition is fixed (Connolly, 2009). However, the geological fluids we simulate here contain solvents (typically H_2O and CO_2) and solutes (volatile and nonvolatile) that may migrate and modify rock compositions along their path (open system). Open system processes that involve changes in environmental conditions (*P*, *T*) and bulk rock chemistry may still be addressed by discretization of the subducting slab (Connolly, 2005; Gorman et al., 2006). This is used here in a simple dissolution-fractionation-transport model. Our fluid dynamics is similar to the model of Gorman et al. (2006), but our thermodynamics contain essential improvements with respect to the nature and composition of the fluid phase. The main novelty here is that all chemical components have a finite solubility in the fluid phase, with mineral solubility relationships conforming to decades of experimental data (see thermodynamic database and comparison with experimental results in supplementary materials). Therefore, while being computationally



Global Su	bduction Input	and Extra	ction Eff	iciency	TIMM1 J	here we we																	
				H_2O			CO_2		~	Ia_2O			$\mathbf{K}_2\mathbf{O}$			SiO_2			Al_2O_3			CaO	
Subductio	in zone LHS	Length (km)	Input (Mt/ yr)	Fore %	Sub %	Input (Mt/ yr)	Fore %	Sub %	Input (Mt/yr)	Fore %	Sub %	Input (Mt/ yr)	Fore %	Sub %e	Input (100 Mt/yr)	Fore %00	Sub %000	Input (100 Mt/yr)	Fore	Sub %~~	Input (100 Mt/yr)	Fore %000	Sub %00
N Pacific	Aleutian	1,500	48.4	20.6	63.9	6.1	3.1	64.5	49.9	3.3	10.2	7.3	-	17.5	15.7	0.4	16.1	3.9	0	8.1	2.6	0.1	26
	Alaska	2,050	65.2	10.6	87.7	5.7	1.9	88.4	70.1	1.1	15.5	13.8	0.1	25.6	22.1	0.1	32.9	5.5	0	19.7	3.5	0	20
E Pacific	N Cascadia	550	11.4	72	91	7	33.6	9.99	11.8	8.7	14.1	2.7	2.5	10.4	3.8	6.5	22.5	0.9	3.1	14.8	0.6	10.5	17.4
	C Cascadia	850	14.8	71.4	91.6	2.5	36.5	9.99	15.8	8.6	14.2	3.3	2.8	11.8	5.1	6.5	22.8	1.3	3.1	14.9	0.8	10.4	17.2
	Mexico	1,700	44.3	69	96.5	4.3	51.1	99.7	54.7	7	17.6	4.4	7.1	61.1	18.5	6.3	23.2	4.6	3.4	13.2	3.2	3.7	5.3
	Guatemala	500	14.4	7.7	96.8	8.6	0.1	37.6	17.5	0.6	10	1.4	0.9	205.6	5.9	0.2	36.3	1.5	0	14.4	1.1	0.1	96
	Nicaragua	275	~	4.3	85.1	3.1	0.1	37.9	10.1	0.5	8.1	0.8	0.6	148.1	3.4	0.2	27.8	0.9	0	6	0.6	0	<i>9.77</i>
	Costa Rica	450	13.7	5.8	93.8	6.6	0.1	35.8	17	0.6	9.6	1.4	0.8	173.4	5.8	0.2	29	1.4	0	9.1	1.1	0	84.9
	Colombia	1,100	26.1	5.5	96.8	10	0.3	53.7	32.6	0.5	11.4	2.6	0.4	154.8	11.1	0.1	39.2	2.8	0	17.4	7	0.1	80.4
	Peru	2,200	62.8	3.8	78.5	23	0.1	40.9	80.3	2.4	13.3	6.4	1.2	168.2	27	0.2	28.6	6.7	0	9.2	4.9	0	67.2
	N Chile	2,000	67.7	2.8	51	8.2	0.5	46.8	85.7	0.4	6.6	7.3	0.3	18.7	28.6	0.1	18.3	7.2	0	13.1	5	0	13.2
	S Chile	2,000	17.1	0	88.1	4.3	0	98.6	25.4	0	11.5	5.4	0	15.9	7.5	0	38.6	1.9	0	28.3	1.2	0	45
W Pacific	Kamchatka	1,100	44	25.1	51.2	3.2	0	14.6	44.3	37.7	41.6	4.5	0.7	11.1	14.7	0.2	8.3	3.6	0	4.5	2.5	0	3.3
	Kurile	1,100	41.4	17.3	45.3	3.4	0	14.2	44.8	0.3	5.1	4.1	0	11.7	15.2	0.2	8.4	3.8	0	4.5	2.6	0	2.5
	Honshu	1,000	47.6	18.6	46.7	5.7	10.7	43.6	49.4	15.7	18.9	5.5	0.9	12.7	16.6	2.9	11	4.1	0	4	2.8	0	21.1
	SW Japan	006	18.6	21.7	91.2	3.1	4.9	98.6	17.7	2.7	16.5	4.4	0.7	18.3	6.2	1.3	40.3	1.5	0.4	29.3	1	6	42.5
	Izu-Bonin	1,300	46.7	9.6	46.7	7.2	0.8	40.6	54.7	8.5	11.4	4.7	0.9	32	18.8	2.7	12.4	4.6	0	4.1	3.3	0	30.5
	Ryukyu	1,000	25.9	3.8	45.7	2.5	0	20.2	32.7	0.6	13.8	2.8	1.2	93.1	11	0.2	11.5	2.8	0	2.5	1.9	0	2.3
	Mariana	1,600	56.5	8.7	96.3	7.4	0.7	97.4	68.1	4	13.1	9	2.1	98.3	23.4	0.2	25	5.7	0	8.8	4	0.2	45.6
	Phillipine	1,000	17.6	14.8	52.1	3.1	1	39.9	19.9	2.3	9.1	7	1.9	60.2	6.5	0.7	12.4	1.6	0	3.6	1.1	0.1	23.3
	Solomon	2,750	113.4	2.9	51.4	15.9	0.4	52.5	146.1	0.4	5.1	12.2	0.5	35	48.7	0.1	11.6	12.1	0	5.3	8.5	0	31
	Tonga	1,500	66.3	3.9	31.9	6.4	0.8	8.4	7.9.7	9.9	16.6	7.6	0.7	38.7	26.7	0.9	6.6	6.6	0	1	4.6	0	1.8
	Kermadec	1,250	34.2	6	44.1	8.7	4.3	26.7	40.9	9.4	12.9	3.8	1.3	21.9	13.7	0.9	9.2	3.4	0	3.5	2.4	0	24.6
Atlantic	Lesser Antilles	850	50.4	57.7	95.5	5.9	8.1	85.5	20.7	17.5	30	9	0.4	9.9	6.8	1	31.3	1.8	0.1	17.9	1	0.2	104.4
	Aegean	1,200	28.6	43.3	88.6	9.6	2.1	22.2	18.3	10.4	20.2	7.1	7	14.1	5.4	2.1	24.3	1.3	0	10.1	0.8	0.5	85
Indian	N Sumatra	1,800	44.9	35.9	85.7	6.7	4.9	78.1	31.5	8	17.6	10.8	0.3	10	9.7	0.2	25.6	2.4	0	14.6	1.5	0	100.2
Ocean	S Sumatra	1,800	61.7	27.4	67.1	4.2	3.9	46.6	52.4	3.8	10.7	12.4	0.9	9.5	16.9	0.8	16.6	4.2	0	8.8	2.7	0.1	11.9
	Java	2,100	74.9	14.5	47.3	8.3	2.5	39.9	79.2	2.3	7.2	11	0.2	10.3	26.7	0.2	9.3	6.6	0	4.2	4.5	0	14.5
Global	Total	37425	1,166.6	18.2	66.3	185.9	3.7	51	1,271.2	5.4	12.9	161.5	1	36.2	421.6	1	18.5	104.7	0.2	6	72.1	0.4	30.9
Note. The IHS and F	values are base HKS models are	provided	upper ma in Tables	ntle hy S5 and	dration s d S6 in S	tate mod	lel (LF ig Info	IS) (ca. rmation	1,166 Mt <mark>S1</mark> . We p	'yr H ₂ C rovide	Nyr glol budgets	al wate at fore-	r input arc (<8	with ca. 0 km de	250 Mt pth) and	yr from sub-arc	mantle (<150]	peridoti km deptl	te). The	full ma	ss balan expresse	ce result: d as a fu	s from nction



efficient, this approach offers greater geological consistency and accuracy in modeling the composition of rocks and fluids in open petrological systems.

We ran a high-resolution variant of the original backcalculation algorithm of Galvez et al. (2015). The core component of all variants of this algorithm, that is, simple- (Galvez et al., 2015), lagged- (Connolly & Galvez, 2018), or micro-backcalculation (this study), relies on finding first the Gibbs energy minimum (equilibrium state at a given P and T) of a system composed of rock and solvent using some appropriate Gibbs energy minimization (GEM) technique, for example the Perplex software here (Connolly, 1990; Galvez et al., 2015). In a second stage, a set of independent chemical potentials derived from this GEM phase is used to estimate (backcalculate) the electrolyte speciation (cf. Galvez et al., 2015 for details). The reason why this basic strategy is so precise in most situations (Connolly & Galvez, 2018) is because the rock and solvents represent the dominant mass (and Gibbs energy) of high-P systems. This is the general rock-dominated criterion which is always realized here. It ensures that the predicted fluid and solid phase assemblage is very close to the Gibbs energy minimum even though the electrolytes are not part of the GEM procedure. A more robust, specific, rock-dominated criterion would require that the mass of every single system's component (e.g., SiO₂, CaO, K₂O, and CO₂ etc.) is dominantly partitioned into the solid phase. Violation of this criterion does occur occasionally, but the impact on our conclusions is marginal. Indeed, without getting rid of both rock-dominated constraints, here we significantly enlarge their domain by equilibrating aliquots of the infiltrating fluid and its solutes sequentially. This ensures minimal fluid (solute)-rock ratio at each iteration, which is nearly equivalent to assuming a constantly low porosity regardless of integrated fluid flux. This is geologically realistic at high P. We then take advantage of the fact that cycles of fluid speciation and fractionation conducted in the rock-dominated limit may lead to profound, and so far overlooked, changes in rock compositions.

In detail, we first discretize a column of rock into grids (resolution provided below), and we assign the bulk composition to each grid according to its position within the four distinct horizontal layers (see Supporting Information S1). We then proceed in three steps: rock-solvent equilibration (by GEM), fluid speciation, and fluid fractionation. In the first step, we compute the chemical potential of the rock and solvent components by equilibration of the bottom grid in the column of rock using the Perple_X software (Connolly, 2005). Table S4 in Supporting Information S1 lists the solid solution models used for this GEM step where melts are not considered. In the second step, the chemical potentials obtained by GEM are used to backcalculate the fluid speciation using a non-platform specific code, which we run on Mathlab® (cf. Galvez et al., 2015). We provide details on the protocol and activity models in the Supplementary Information S1. In the third step, the fluid (solvent and solutes) is fractionated from the bulk composition of the local grid, and added to the bulk composition of the grid immediately above before repeating the process. For enhanced thermodynamic consistency and chemical accuracy, we estimate fluid speciations in high-resolution mode (i.e., by microbackcalculation) where aliquots of fluids (as opposed to the bulk fluid) are equilibrated, speciated, and fractionated successively. Discretization of the fluid volume (set here to n = 10 sub-volumes) can be made as dense as desired, at the cost of computation time. This mode converges to the rock/fluid composition asymptotically, unless a rock component is entirely dissolved (i.e., fluid-dominated system), that is, when the amount of a dissolved component is predicted to be higher in the fluid than the leftover in the rock. In this case, only the leftover mass of this specific rock component is allowed to dissolve in the fluid. This is done under mass and charge balance constraints, that is, the dissolved component traverses conservatively the rock grid in which it is temporarily out-of-equilibrium. This minor rule ensures mass conservation during the entire transport and subduction sequence.

Complete fluid expulsion assumes high permeability within the reactive zone in the slab (Ague, 2003). Our model implicitly assumes vertical fluid transport within the slab as driven by buoyancy force. Although Mottl et al. (2004) supported this model at Mariana fore-arc, other studies (e.g., Angiboust et al., 2014; Galvez et al., 2013; Kawano et al., 2011) show that interface-parallel fluid flux is important in subduction settings, suggesting our model may overestimate vertical fluxes. The dissolution-fractionation-transport procedures are repeated until a depth of 220 km is reached. The grid resolution along in-slab direction is ~20 m within the thin sedimentary and basaltic layers and increases to ca. 200 m within the thicker gabbroic crust and AUM layers. Subsequently, the entire column of rock is moved downward according to the thermal model considered. The descending rate of slabs sets the subducted mass flux of rock columns (Syracuse et al., 2010). We have chosen an increment of ca. 1 km/step to move down the slab from 20 km depth until 220 km depth, as preliminary tests have shown that



this discretization optimizes computation time while offering sufficient resolution to monitor the mass transport process. It limits overstepping of the dehydration reactions, and ensures minimal fluid/rock ratio.

Temperature at each grid point is interpolated based on the thermal model from Syracuse et al. (2010). Hotter alternatives (Penniston-Dorland et al., 2015) are tested (Figures S21 and S22 in Supporting Information S1) and discussed. Pressure is related to depth assuming lithostatic pressure gradient. The *P*-*T* field is therefore specific for each slab.

3. Results

3.1. Devolatilization Pattern of Cascadia Versus Magnetotelluric Data

Figures 1c and 1d show a match between fluid release events—associated with the carbonate sediment at ca. 40 km, basalt layer at 50–80 km, and with the serpentinized mantle at 90–100 km, respectively (cf. Figure S2 in Supporting Information S1 for details on fluid flux and composition)—and the depth of a low resistivity area for the Cascadia subduction zone (McGary et al., 2014). The water plume linked to upper mantle dehydration matches an extensive zone of low resistivity located between 80 and 110 km depth. A minor fluid plume, sourced within the sedimentary layer at ca. 40 km, is also qualitatively consistent with a weak resistivity response observed in this depth interval. This match is taken as a qualitative validation of our subduction model setup.

3.2. Interslab Variability of Element Release Out of Slab: H₂O

The location of slabs dehydration events varies dramatically across different subduction zones (Figure 2 and Figure S2 in Supporting Information S1). For example, the main dehydration event of the Nicaragua subduction zone occurs along a rather broad depth range between 125 and 160 km. The serpentine/chlorite dehydration reactions occur first in the cold and low P top of the AUM, which creates a spatially broad fluid plume rising across the mafic and sediment layers (Figure 2b). This pattern differs in the warm slab of Cascadia, for example, where chlorite destabilizes first in the hot and deeper segment of the AUM, and spatially disconnects it from the serpentine dehydration front (very hydrated top of the AUM, Figure 2a). Eclogitization of the sediment and basalt releases a free fluid phase between 40 and 60 km. The entire Nicaragua slab is dehydrated by 170 km depth, and the entire Cascadia slab is dehydrated by 100 km depth. By contrast, only the top 500-1,000 m of the cold Honshu oceanic mantle (antigorite) may dehydrate at a depth of about 200 km. The water released from the AUM is transported to the igneous layer. Most of the slab water input, locked in chlorite and serpentine (or Phase-A at deeper depth) in the upper mantle or in phengite and lawsonite in the igneous layer (Figure 2), is predicted to bypass the sub-arc region. Globally, about 212-218 Mt H₂O/yr is released to fore-arc depth (20-80 km), and this value is largely independent of the hydration state of the AUM (see Table 1, and also Table 2 for LHS, Table S5 in Supporting Information S1 for IHS, and Table S6 in Supporting Information S1 for HHS models). By contrast, the water released within the 80-150 km depth shows marked sensitivity to the hydration state of the AUM (Table 1). Between 561 Mt/yr ($\eta_{H20}^{\text{subarc}} = 48.1\%$ of trench input, LHS, Table 1) and 1,015 Mt/y ($\eta_{H20}^{\text{subarc}} = 41.8\%$ of trench input, HHS) is released in the depth interval.

3.3. Interslab Variability of Element Release Out of Slab: Carbon

Carbon is present as aragonite, dolomite, and magnesite in the sediment layer of our model (Figure 2). Figures 3 and 4 show that carbon is released from the rocks in pulses, regardless of the hydration state. For all the slabs, the location of carbon pulses is systematically associated with dehydration events (e.g., Figures 2 and 3). For example, the carbon load from the Cascadia sedimentary layer is released within a narrow depth range (60–80 km, Figures 3a-3d) associated with eclogitization of the sediment and igneous layers and infiltration of H₂O from the AUM flushing C from reaction sites (Figures 3a-3d).

The AUM hydration state does not modify the pulsed nature of carbon release (Figures 3 and 4). However, key quantitative aspects of this release, for example, its rate and magnitude are sensitive to the hydration state (Figures 5 and 6b). For example, the integrated decarbonation flux out of the Nicaragua slab increases by 20%–30% from LHS to HHS models (Figures 4b and 6b). Because the carbon output is sensitive to the fluid expulsion (transport) from the AUM, we classify this regime as *transport* (water) *limited* (Figure 6 and Figure S4 in Supporting Information S1). This is not the case for the hot slabs such as Cascadia and Mexico, where the



a. Cascadia



Figure 2. Relative water mass variations per 100 g of rock (H_2O mass subtracted by the initial H_2O mass divided by the total rock mass × 100). Each phase field present the associated equilibrium phase assemblages within a hot (North Cascadia, Western U.S.), intermediate (Nicaragua, Central America), and cold (Honshu, Japan) subduction end-members (HHS model). Red color indicates net gain in mass due to metasomatism and blue color indicates net loss. The in-slab depth axis (*y*-axis) is slightly tilted to better visualize the subduction slab stratification. The initial water input in the mantle is assumed to vary linearly along the in-slab direction, which explains the drier phase assemblages in the lower part of the ultramafic section (cf. Figures 3a–3c). The thickness of the sediment is plotted as 1 km for Nicaragua and Honshu so that the phase assemblages can be seen. Detailed flux of water and composition of fluid along the slab surface (Figure S2 in Supporting Information S1) and composition of fluid along selected vertical transects of the three systems (Figure S3 in Supporting Information S1) are provided in the supplementary material.

decarbonation efficiency (Figures 4 and 5 and Table 2) is invariably close to 100% regardless of the hydration state. Because the carbon output is controlled by the initial carbon supply, we classify those slabs as supply limited (Figures 6 and S4 in Supporting Information S1). Interestingly, cold slabs (e.g., most slabs in the western





Figure 3.





Figure 4. Cumulated fraction of CO_2 lost from top of the slab for the three subduction zones, highlighting contrasted sensitivity to altered upper mantle hydration state for the carbon limited (Cascadia), water limited (Nicaragua), and frozen (Honshu) regimes.

Pacific such as Honshu, Ryukyu, Izu-Bonin, Tonga, and Java, Figure 5), are also insensitive to hydration states, but for different reasons. Figures 6c and 6d show that those slabs release negligible amounts of H_2O from the AUM before sub-arc depth, regardless of the hydration model, and therefore retain a large fraction of their carbon load to beyond sub-arc depth as magnesite. The feature is therefore primarily caused by the hydrodynamic regime of the slab, that is, the lack of water to mediate the dissolution, and only proximally to the slab's thermal structure. We call this regime thermally *frozen*.

The provenance of H₂O available for carbon dissolution can vary widely (Figure 7). For example, AUM dehydration represents less than 20% of the H₂O released from the warm slab of Northern Cascadia before sub-arc depth (LHS), but it contributes for 60% of its C released (Figures 7a and 6b). Conversely, AUM dehydration (30% slab H₂O) contributes for less than 10% of the carbon released from the South Chile slab (Figure 7c). There, the C dissolution is mostly driven by water originating in the sediments and igneous layers: 60% of slab H₂O (sediment + igneous layers) removes 90% of slab carbon. Globally, the fraction of trench carbon released at forearc depth (Table 2) is $\eta_{\rm CO2}^{\rm forearc} = 3.7\%$ (ca. 7 Mt CO₂/yr). However, placing the forearc-subarc threshold at a depth of 90 km brings this value to $\eta_{\rm CO2}^{\rm forearc} = 17\%$ (32 Mt CO₂/yr). The flux is mostly driven by devolatilization of the igneous layer (Figure 2). The CO₂ extraction efficiency to subarcdepth is constrained between $\eta_{\rm CO2}^{\rm subarc}$ = 51.0% (LHS model, 94.8 Mt CO₂/yr or 25.9 Mt C/yr) and $\eta_{\rm CO2}^{\rm subarc}$ = 61.9% (HHS model, 115 Mt CO₂/yr or 31.4 Mt C/yr), corresponding to $\sigma_{CO2}^{\text{subarc}} = 49.0\%$ (LHS) and $\sigma_{CO2}^{\text{subarc}} = 38.1$ (HHS).

Overall, AUM dehydration contributes only ca. 25% of global slab C loss before sub-arc (Figure 7d), and therefore the majority of the subarc decarbonation is predicted to be driven by H₂O originated from the sediment and

igneous crust. About 40% of the sedimentary and igneous carbon remains after sub-arc depth, while about 80% of the ultramafic C remains (Figure S5 in Supporting Information S1). In other words, our current model design predicts inefficient slab decarbonation, and this inefficiency affects all layers, including the most heavily flushed sediments and igneous layers. Of course, AUM dehydration does promote continued dissolution of the slabs (Figures 3 and 7d) beyond 150 km depths. This flux may flow toward the back-arc region, unless it is channeled back toward the sub-arc laterally.

3.4. Interslab Variability of Element Release Out of Slab: Nonvolatile Elements

Our qualitative findings above apply to all other rock forming elements. Most notably, the release of K, Na, Ca, Mg, Al, and Si occurs in discrete pulses that are controlled by the sequence of dehydration reactions within the slabs (Figure 2). The composition of fluids available for mantle wedge metasomatism is generally rich in Na, Si, C, and Ca (ca. 0.1 to 1 molal beyond 80 km, Figures 3a and 3i), poor in K until ca. 90 km depth, reflecting the stability of white mica. They differ greatly by their Al content (Figure S2 in Supporting Information S1). It is highest at high-T (which varies from slab to slabs) where the Al reaches the top of the slab in the form of soluble alumino-silicate complex, and lowest at lower T where Al is highly insoluble. Unlike C and Si, the hotspots of Al release to sub-arc depth are the trenches of Mexico, South West Japan, or Alaska, broadly reflecting their compositional enrichment in Al. At higher P and T, the metasomatic fluids become more potassic while the slabs, taken as whole, get progressively depleted of this element (Figures 3 and 9, Figure S2 in Supporting Information S1). An interesting feature of the Nicaragua sediment is that it is K free after 90 km depth (Figure 3), and yet the fluid

Figure 3. Pattern of water, carbon, Al_2O_3 , and K_2O redistribution and release for warm, intermediate, and cold slabs as a function of hydration states. Relative H_2O , CO_2 , Al_2O_3 , and K_2O mass changes (Δm_i) of components are given in % of the total rock mass at first grid. The percentage values displayed as inset within the panels express this variation in % of the input mass of the component at first grid. The warm color indicates absolute mass gain and cold color indicates absolute mass loss. For Honshu slab, the first grid undergoes dehydration in the sediment because of its excessively high initial H_2O content. For illustration purpose, we compare the mass variation with respect to the state after the fractionation of the first grid.





Figure 5. Total top-of-the-slab CO₂ flux (left panel) and decarbonation efficiency (right panel, η CO₂) to sub-arc depth 150 km for all the studied subduction zones, ranked geographically. The left panel shows the integrated flux of CO₂ in Mt/yr from surface to sub-arc depth (150 km). It shows that the canonical value of ca. η CO₂ = 0.5 (Volk, 1989) is only relevant in average, and today (Galvez & Pubellier, 2019). Note that the E-Pacific slabs have the highest extraction efficiency. The decarbonation efficiency of slabs generally increases with the geothermal gradient (Figure S4 in Supporting Information S1).

crossing it and reaching the mantle does contain K inherited from the mafic (not sedimentary) layers (Figures S2 and S3 in Supporting Information S1). This is a peculiar case of partial equilibrium between the fluid and the rock. For all the subduction zones studied, the fluids that reach the top of the slab are much richer in Al, Si, C and much poorer in Mg and Fe than they are at the site of their production (in the mafic and ultramafic sections, Figure S3 in Supporting Information S1).

The sensitivity of mass fluxes to slab hydration state differs between oxides and between slabs (Figures 3, 8 and 9) For example, the hot slabs of Central Cascadia and Mexico were supply limited for C because this element was in sufficiently low concentration to be completely removed from the sediment by sub-arc depth (Figure 6). However, the extent of their desilicification and dealuminification (rich in SiO₂, Al₂O₃) does prove sensitive to the AUM hydration state, which makes them transport limited for those components. By contrast, the desilicification and dealuminification of slabs such as Costa Rica and Guatemala depends heavily on AUM hydration (transport limited), just as their decarbonation did (Figures 6 and 8). A last salient feature is that individual slab units (sediment, basalts) may fall into a specific transport regime, while the slab as a whole may behave differently. For example, the Nicaragua sedimentary layer, and that of other carbonate-rich subduction zones, are supply limited with respect to K (i.e., sediment loses all its K by sub-arc depth, Figure 9n). However, the release of K from the slab as a whole shows some dependence to hydration state because K is also distributed within the igneous layer.

3.5. Intra-Slab Redistribution of Elements

Our model also provides insights into patterns of element redistribution within the slab (intra-slab metasomatism), which contributes to significantly modify the rock compositions. However, the concentrations of components in a rock are by themselves a poor indicator of metasomatic transport (Aitchison, 1982; Gresens, 1967). This is because the concentration changes of a component (Δc_i) hinges on the behavior of the other rock components as well. For example, the concentration of a nonvolatile component may be residually enriched when H₂O and CO₂ are massively lost from the system. Here, we use the relative change in mass of a component scaled by its initial input Δm_i as a primary metric to assess the metasomatic redistribution of nonvolatile components throughout the slab (Figure 9). We found that most mass changes occur at lithological boundaries, where contrasts in chemical potentials between adjacent lithologies are the largest, in particular between the sediments and the igneous layers (Figure 3). Another salient feature is that element concentration (Δc_i) and mass (Δm_i) variations may vary in opposite directions (Figure 9, and exhaustive data set in Figures S6–S12 in Supporting Information S1).

For example, Δc_{SiO2} (increases) and Δm_{SiO2} (declines) are anticorrelated in the bottom of the sedimentary layers of Nicaragua, Cascadia and Honshu, which are affected by large volatile loss. Similar anticorrelated trends for Al₂O₃ (most slabs, Figure S7 in Supporting Information S1) and for K₂O variations in Honshu (NE Japan), Aegean and North Sumatra (Figure S10 in Supporting Information S1) are caused by net loss of H₂O and CO₂ from the rock, which counters the net gain in mass of the nonvolatile component. This primarily reflects residual change (Figure 9). Conversely, Δm and Δc may covary perfectly in the case of H₂O, CO₂, Al₂O₃, and K₂O. For example, most carbonate-rich sediments (e.g., Nicaragua, Costa Rica, Peru, Columbia, and Guatemala, cf. Supporting Information S1 for exhaustive data set in Figure S10) lose all of their initially low K₂O (Δm and Δc decline), whereas they accumulate Al and Si in all or parts of their thickness (Δm and Δc rise). This feature is mostly associated with infiltration of fluids produced in the dehydrating AUM, which reach the top of the slab at higher temperatures. In this case, dissolved Al promotes the growth of grossular garnet in the sediments (details in discussion). Other slab sediments, such as those of North Chile ($\Delta m_{K2O} = +20\%$), Kurile, and Sumatra slightly increase their K mass with depth (Figure S10 in Supporting Information S1), while those richer in Si and Al display net Si and Al mass loss (Figures S6 and S7 in Supporting Information S1). The link between composition





Figure 6. Panels (a and c) show the cumulated CO_2 and H_2O output per km of trench from surface to 150 km depth. Panels (b and d) show the flux difference between high hydration state (HHS) and low hydration state (LHS) model. Slabs that are sensitive to the hydration state of AUM are labeled in black (water-limited regime), slabs that are insensitive because they are fully decarbonated with LHS model are labeled in red (carbon-limited regime), slabs that are insensitive because they do not lose much C even with HHS model are labeled in blue (frozen regime). The ranking in (d) (H₂O) follows that in (b) (CO₂), so that the three regimes are separated into three regions. For example, in Cascadia, although the water flux difference between HHS and LHS models is high as shown in (d), the CO₂ flux difference is rather low as shown in (b), indicating that carbon is depleted in this slab and extra water will not increase the CO₂ flux further. Note that the forearc flux are particularly sensitive to the choice, semi-arbitrary, of the fore-arc/sub-arc threshold. Here it is placed at 80 km (see text for discussion).

and patterns of mass variations by metasomatism is also observed for Ca. Ca-deficient sediments accumulate Ca, and the effect is dramatic for Mexico (+120%), Ryukyu (+180%), and Kurile (200%) (Figure S8 in Supporting Information S1). By contrast, the lower section of limestone-subducting slabs loses about 50% or more of its Ca content by 150 km depth (Figure 9). This effect is not observed at the top of this sedimentary layer. This effect may be caused by insufficient H_2O infiltration.

We also found that patterns of mass changes may not be monotonous, and the profile of CaO in the Honshu sediment is a striking example. After the carbonate (dolomite and aragonite) is fully dissolved from the sediment by 110 km depth, a process accompanied by net Ca loss (Figure 9t), Ca accumulates in the carbonate-free sediment and is marked by metasomatic garnet growth.





Figure 7. η - η diagrams showing the decarbonation efficiency η CO₂ versus dehydration efficiency η H₂O for (a and b) Cascadia, (c) South Chile, and (d) the global subduction system (results from model low hydration state [LHS]). Curves, or portion of curves, with highly positive slopes indicate that most of the carbon is removed with minimal H₂O release from the slab (i.e., high C solubility), while portions of curves with flat trends express the opposite. The arrow in (a and b) indicates the location where mantle dehydration starts. (d) A near-linear trend is observed for the global subduction system. The η - η diagrams for the other slabs can be found in supplementary materials (Figure S19 in Supporting Information S1). Panel b (red) is the result from runs where H₂O content in the upper mantle of North Cascadia (British Columbia) is 7.5 times lower than even our driest LHS model, due to limited plate bending and H₂O penetration within the oceanic mantle (cf. Canales et al., 2017 for details), or 1.5 wetter than our LHS model (blue). Regardless of the hydration state, the similarity shows that the slab is supply limited.

Moreover, most slab sediments lose Na due to the Na-deficient character of our ultramafic or igneous fluids, with Δm_{Na2O} variations ranging from a few percent (e.g., Aegean) to almost 100% (e.g., Ryukyu, Mexico) at subarc depth. There are a few exceptions, but in those cases the accumulations never exceed a few relative percent at most (Figure S9 in Supporting Information S1). Mg only accumulates at the base of the gabbro layer after dehydration of the upper mantle (less than a few relative %, Figure S18 in Supporting Information S1 for Nicaragua), and to a lesser extent at the bottom of Mg-depleted sediments. We expect this effect to be more pronounced in case of direct ultramafic/sediment interfaces, which are not simulated here. This intermediate storage tends to lock Mg within the slab, instead of releasing it to the mantle wedge.





Figure 8. Global ranking of the SiO₂, Al_2O_3 , Na_2O , and K_2O fluxes for subarc and forearc depth with the same setup as in Figure 5. Total fluxes and total flux per km of trench are given.

3.6. Metasomatic Control on Carbonate Dissolution

Metasomatic mass redistribution affects the release of C from the slab (Figure 9). This effect is monitored by computing the system's chemical potential difference $(\Delta^{s}\mu)$ between two models, one where all components are mobile (μ^{spec}) , one where only molecular CO₂ and H₂O are mobile $(\mu^{nonspec})$, hence $\Delta^{s}\mu = \mu^{spec} - \mu^{nonspec}$. The superscript *s* indicates that the symbol Δ does not denote a temporal or spatial variation, but a contrast between speciation models. The dissolution of a carbonate (e.g., CaCO₃) in a fluid can be described by an equilibrium relating the Ca-carbonate constituent to its thermodynamic components (e.g., CaO and CO₂), regardless of their hosts (fluid or solid) and of the silicate environment:

$$CaCO_3 = CaO + CO_2 \tag{3}$$





Figure 9. The panels (a, i, and q) show the flux of oxides out of the slab surface. The other panels show the concentration change and mass change of the oxides at the bottom of the sediment layer, where the metasomatic effect is strongest. Relative concentration (Δc_i) and mass variations (Δm_i) of component *i* are given in % of the input mass of the component at first grid. Unlike concentration variation, the mass variation reflects the actual metasomatic effect. Models assuming infinite compatibility of nonvolatile major elements in the hydrous fluid phase would typically show no net mass change (i.e., flat blue curve at 0% mass change for nonvolatile components). Concentration changes may always occur due to residual enrichment or dilution when volatiles are lost or added, respectively. Only the results of the low hydration state model are displayed.

Under thermodynamic equilibrium of a closed system, Equation 1 gives:

$$\mu(\text{CaCO}_3) = \mu(\text{CaO}) + \mu(\text{CO}_2) \tag{4}$$

which also implies that the two potentials μ (CaO) and μ (CO₂) cannot vary independently if a pure CaCO₃ endmember (i.e., aragonite) is stable, regardless of the system's composition. For example, if μ (CaO) declines (i.e., higher compatibility of CaO in the rock and lower solubility in the fluid), μ (CO₂) rises to offset this decline and maintain the equality, which practically means greater CO₂ concentration in the fluid if the activity coefficient remains similar. We do observe that $\Delta^{s}\mu$ CaO and $\Delta^{s}\mu$ CO₂ are strictly anticorrelated along the subduction path (Figure 10a), indicating that aragonite is indeed always stable in the sedimentary mineral assemblage of Nicaragua (see also Figure 2). And the marked deviation of $\Delta^{s}\mu$ CO₂ from the non-speciated reference frame ($\Delta^{s}\mu$ CO₂ = 0) is the most direct marker of non-volatile metasomatism-the distinctive feature of our model. As could be expected from a metasomatic effect, the fluctuations of $\Delta^{s}\mu$ CaO track the two main fluid infiltration events into the sediment, at 80 and 140 km (Figures 2b and 10b). These events contribute not only mafic/ultramafic





Figure 10. (a and c) Chemical potential difference for various components between our speciation model (μ^{spec}) and a non-speciation model (only CO₂ and H₂O are mobile, and other oxides are infinitely compatible with rocks, μ^{nonspec}) at the bottom of the sediment layer in Nicaragua. The chemical potentials of CaO and CO₂ are anti-correlated, implying stability of aragonite. The chemical potential difference between those two models is a marker of metasomatism. (b) Mass flux of dissolved carbon through the bottom of the sediment layer (normalized to CO₂ mass) and normalized fluxes of SiO₂ and Al₂O₃. (d) Garnet, aragonite and phengite mode percentage. Note the correlation of garnet growth with fluid pulse.

 CO_2 and H_2O (Figure 2b), but also CaO, Al_2O_3 and SiO_2 into the system (Figure 10b, see also Figures 9i–9p). The energetic deviations between the speciation and non-speciation model for each component may sometimes reach several thousands of joules, especially when elements are particularly soluble (e.g. K, Na) or particularly tied to the fate of the alkalis (e.g. Al). The silica component is an exception, however, as the deviation of $\Delta^s \mu SiO_2$ for all sediments is generally minimal. This means that the net silicification (or desilicification, N/S Chile, SW Japan) of sediments (Figures 9j and S6 in Supporting Information S1) is insufficient to affect the overall energetics of the SiO₂ component in those systems.

4. Discussion

4.1. Geological Applications

4.1.1. Return Flux of H₂O and Comparison With Ocean Depth Variations

Our predicted devolatilization events for the Central Cascadia subduction zone match the magnetotelluric record of low mantle resistivity (Figure 1). This supports a flux-melting origin for the resistivity anomaly at the mantle source of Cascadia arc magmatism, even though the water fluxes involved may be limited (Canales et al., 2017). The limited fore-arc devolatilization that we predict globally, and in the Kamchatka subduction zone in particular, is consistent with findings by Konrad-Schmolke et al. (2016). However, our results also show fore-arc estimates are surprisingly sensitive to model design. For example, shifting the fore-arc/subarc threshold to 90 km depth doubles the predicted forearc H_2O flux (Figure S13 in Supporting Information S1). This could typically account for interface-parallel upward fluid transfer, in which a fraction of the fluid pulse released from the basaltic layer between 80 and 90 km would be channelized to the fore-arc by migration within the slab, or along the subduction interface (Angiboust, Glodny, et al., 2021; John et al., 2012; Kawano et al., 2011).



Only our IHS and LHS models (Table 1) yield sub-solidus H₂O releases to sub-arc depth that match observational constraints on global H₂O degassing at volcanic arcs, that is, ~650 Mt H₂O/yr (Fischer, 2008). Our corresponding return flux of H₂O to the mantle (393 Mt H₂O/yr for LHS, and 571 Mt H₂O/yr for HHS) imply ca. 100 and 300 m sea-level decrease over the Phanerozoic, respectively (Table 1). This is within the acceptable range of sea-level variations permissible based on observational constraints, that is, 0-360 m (Parai & Mukhopadhyay, 2012). The global return flux of our HHS model (1,195 Mt H₂O/yr) would imply excessive sea-level variations meaning it is not realistic globally. But it may be relevant regionally when slabs are particularly cold (e.g., Mariana, Honshu; Cai et al., 2018). The impact on C and nonvolatile element fluxes would be sensible only if the AUM would dehydrate to some significant extent before sub-arc depth, which is not observed here (Figure 2c). But the contention by Penniston-Dorland et al. (2015) that the thermal models of Syracuse et al. (2010) do not reflect accurately the metamorphic record leaves room for debate. It is important to note, however, that the metamorphic record studied by the former includes rocks detached from the slab, underplated and exhumed, i.e. geodynamic contexts quite different from the subduction zone structures simulated by Syracuse et al. (2010). Both studies may therefore be relevant in different geodynamic contexts. In the present context of a mechanically continuous slab, our work suggests that the low upper-mantle hydration states of Van Keken (2011) or Schmidt and Poli (2003), and even the intermediate hydration state of Hacker (2008), are all compatible with the constraint of limited sealevel variability in the phanerozoic if one considers that sub-arc depths extend to 150 km at least (as opposed to ca. 100 km in Parai & Mukhopadhyay, 2012).

4.1.2. Regimes of Carbonate Subduction

Another implication of our work concerns the fate of carbonate. Most models of carbonate subduction to date (Gorman et al., 2006; Kerrick and Connolly, 1998, 2001a, 2001b) support their relatively high thermodynamic stability in subducting slabs. Predicted decarbonation efficiencies range from 40% in Gorman et al. (2006) to 18%-70% in Johnston et al. (2011), and this is qualitatively supported by field observations (Collins et al., 2015; Cook-Kollars et al., 2014; Lefeuvre et al., 2020). But large uncertainties in slab parameterizations and carbon solubilities have led others to suggest that most slab C may be dissolved, released, and stored temporarily in the fore-arc and mantle wedge (e.g., Barry et al., 2019; Kelemen & Manning, 2015). Our work helps to clarify this conundrum. We confirm that C solubilities in metamorphic fluids may have been locally underestimated in previous models of slab decarbonation (Gorce et al., 2019; Gorman et al., 2006; Kerrick and Connolly, 1998, 2001a, 2001b; Tian et al., 2019). By comparing the C concentration and Ca/C ratio of fluids predicted by speciated (hybrid electrolytic/molecular) and non-speciated (molecular) models, Galvez et al. (2016) have shown that this discrepancy in C concentration may reach at least 2 orders of magnitudes at $T < 400-500^{\circ}$ C in pelitic or basaltic lithologies because the dissolution of carbonate may be mostly congruent at those temperatures (Caciagli & Manning, 2003; Fein & Walther, 1987). However, fluid fluxes are also vanishingly low at those temperatures (Figures 1d and 2) such that the effect of enhanced CO_2 solubility is countered and turns out inconsequential in the predicted slab carbon budget, by 150 km depth at least. In other words, if one assumes carbon is part of a cohesive slab, and fluxes are dominantly vertical, then our results do support the existing notion that about half of subducted C (ca. 25 to 31 Mt C/yr, 50%–62% of C input) may be released before sub-arc depth (Caldeira, 1991; Gorman et al., 2006; Hoareau et al., 2015; Volk, 1989), and the other half subducted beyond (Figures 6a and 7d).

But this does not entirely solve the above conundrum as we noted that our fore-arc H_2O and mass fluxes may be underestimated by a factor of 4 at least if the hydrodynamic regime is only slightly modified, ie. if lateral transport of fluid over 10 km is considered. Globally, this is a shift from 7 to 32 Mt CO₂/yr at fore-arc, or 17% of the input (Figure S13 in Supporting Information S1). This sensitivity may be quite dramatic locally depending on slab composition and temperature profile. For example, our predicted fore-arc C flux in Costa Rica is negligible, 0.0015 Mt CO₂/yr (Figure 6a and Table 1), which matches the lower-end isotope-based estimate of Barry et al. (2019), who report a flux of 0.005–0.57 Mt CO₂/yr in this region. But if one allows a fraction of the fluids released at slightly greater depths (e.g., between 80 and 90 km) to be channelized from the sub-arc to the forearc by migration along the subduction interface (e.g., Angiboust et al., 2014; Kawano et al., 2011), or along the impermeable carbonate sediment and its mafic (or ultramafic [e.g., Galvez et al., 2013]) basement, this estimate rises to 0.4 Mt CO₂/yr. This happens to match the upper estimate of Barry et al. (2019) for Costa-Rica. Although this match may well be fortuitous, the sensitivity we report shows that it will be difficult to draw a definitive mass-balance estimate for the forearc, ie. one of the most dynamic and compositionally complex environment of the subduction zone. Therefore, our study supports the contention by Kelemen and Manning (2015) and Barry et al. (2019) that carbon release and storage toward the forearc may be greater than previously thought, as well as the notion of a high thermodynamic stability of carbonate beyond forearc and to subarc depth at least (Collins et al., 2015; Cook-Kollars et al., 2014; Gorman et al., 2006; Lefeuvre et al., 2020). The complex hydrodynamic regime of fluids in the slab may be the primary cause for this conundrum, more than the existence of possible inaccuracies in thermodynamic models of subduction zones.

Plank and Manning (2019) and others (Galvez & Pubellier, 2019) have stressed the importance of compositional heterogeneities between subduction zones. Our results reinforce this notion, as the decarbonation efficiencies of the western Pacific slab is much lower than that in the Eastern Pacific. This difference reflects fundamental compositional and thermal controls. Because subduction efficiencies are typically used as a parameter in long-term climate models such as those of Hoareau et al. (2015); Volk (1989), our work stresses that it should be treated as a variable, in space and certainly in time as well. Caution is therefore needed in applying numeric values of subduction efficiencies to situations where tectonic configurations, sediment composition, and/or slab geometries may have differed from today (e.g., before the Mesozoic).

Our global output from the slabs is lower than field estimate of passive CO_2 degassing at volcanic arcs of 55 Mt/ yr by Fischer et al. (2019), but the latter may include a fraction released by thermal metamorphism and assimilation of carbonate accreted (with both pelagic and shelf origins) to the overriding crust (Johnston et al., 2011; Lee et al., 2013; Mason et al., 2017). More importantly, our sensitivity test for hydration state revealed *transport* and *supply* limited regimes of element subductions, as well as thermally *frozen* slabs (Figure S4 in Supporting Information S1) which informs future research targets. For example, subduction zones of intermediate thermal structure are transport-limited, that is, C fluxes are sensitive to the hydration state of the AUM (Figure 5). This regime includes most slabs of the Sumatra, eastern Pacific, and Aegean (see Table 1 and Figure 5). This is where most C is released (Costa Rica and Guatemala are the largest C emitter per km of trench; Figure 6). We therefore need a more accurate knowledge of the hydration state of those specific slabs. Subduction zones that are hot, such as Mexico and Cascadia, are supply (C)-limited that is, C fluxes are not sensitive to the hydration state of the AUM simply because that the slabs are depleted of carbon at shallow depth. In this case, priority should be given to improving knowledge of their carbon content and distribution. Most of the subduction zones of the western Pacific (e.g., Mariana, Izu, Tonga) also do not depend on the AUM hydration state. In that case, the absence of AUM dehydration before sub-arc depth prevents C dissolution from those slabs which are coined *frozen*.

4.1.3. Metasomatic Redistribution of Elements and Arc Magma Signatures

Knowledge of the absolute mass change of an element typically matters in interpreting the chemical budget and isotopic signature of a rock (Angiboust et al., 2014; Bebout, 2007; Bebout & Barton, 1993; Galvez et al., 2013; Miller et al., 2009; Penniston-Dorland et al., 2012; Piccoli et al., 2016; Vitale Brovarone & Beyssac, 2014), whereas concentration changes typically matter for deriving the geophysical properties of a rock, including its rheology, mechanical behavior, and seismic properties (e.g., Connolly, 2005). Isolating both may be cumbersome, as it is well-known that compositional data may not reliably reflect metasomatic mass transfers (Ague, 2011; Aitchison, 1982; Gresens, 1967). Our results provide a first numerical demonstration that this phenomenon is at play in subducting slabs. We find that patterns of mass and concentration changes may be anticorrelated or correlated, reflecting the antagonistic role of metasomatism versus residual (passive) enrichment caused by volatile loss. By monitoring mass and concentration changes simultaneously, we confirm the possibility of early slab depotassification and dehydration as reported by Connolly and Galvez (2018), but this seems to only apply to certain slab compositions. The effect is most dramatic for carbonate rich and H₂O deficient slabs (e.g., Nicaragua, Costa Rica), while some clay-rich slab sediments are net accumulator of alkali, for example, Alaska and Chile.

More importantly, all chemical elements are redistributed within the slabs before being released from them, and this redistribution reflects local gradients in chemical potential. As a result, most compositional and mass changes occur at the sediment/basalt and basalt/ultramafic interfaces (Figure 3). We note that our 4-layer slab structure is most representative of fast-spreading "pacific-type" subduction settings. Atlantic-type environments that lack a continuous mafic layer are relevant as well (Agard, 2021) and they would include highly contrasted sediment/ serpentinite interfaces (e.g., Galvez et al., 2013; Malvoisin et al., 2012; Scambelluri et al., 2022) which are not represented here. Interestingly, the first fluids produced in the altered ultramafic and mafic layers may be rich in both K and Mg, ie. "shoshonitic". This is due to the highly incompatible nature of K, a minor component in those lithologies (Figure S3 in Supporting Information S1). However, regardless of where the fluids are produced in the rock column (e.g., basalt or ultramafic layers), those have generally acquired a clear sedimentary character by the time they reach the top of the slab. They have become enriched in Si, Al, alkali, carbon, and orders of

magnitude poorer in Mg than ultramafic fluids (Figure S3 in Supporting Information S1). In other words, we have demonstrated that slabs operate as "chromatographic" columns, and this effect should be considered in discussing the role that slab fluids may play in modifying the composition of the mantle wedge through time.

The fluids reaching the slab/mantle wedge interface are enriched in C, Si and Na (cf. Figures S2 and S3 in Supporting Information S1 for details). However, only hot fluids deliver Al in appreciable quantity to the slab surface, that is, at different depths depending on slabs thermal structure. This not only supports a metasomatic origin for jadeites (Angiboust, Muñoz-Montecinos, et al., 2021; Harlow & Sorensen, 2005), but it suggests that the alumina content (e.g., their aluminum saturation index) of subduction zone fluids may be used as chemo/geothermometer. The experimental evidence for the high incompatibility of K in high-*P/T* metasediments (Schmidt, 2015; Schmidt et al., 2004) suggests that the mobility of this element may be elevated at depth. The increasing K/Na ratio of our fluids with depth and temperature supports this notion (Figure S2 in Supporting Information S1). But this incompatibility also implies that dry segments of the slab maintaining K, Na in their compositions may be metastable with respect to a fluid phase. For example, we attribute the flat profile of sodium in the Cascadia sediment after 100 km (Figure 9e) to this metastability effect, caused by the lack of water influx. This may, in turn, cause extreme fluid peralkalinity if a fluid traverses the rock later in the subduction process.

Metasomatic decarbonation reactions have been described in the field (Abart, 1995; Ague, 2003; Bucher, 1999; Ferry et al., 2011), but never tested quantitatively in computational models of slabs compositions. We found that the dissolution of carbonate in the sediment of Nicaragua (and e.g., Tonga, Costa Rica, and Mariana) is associated with a synchronous gain in masses of Al_2O_3 (Figure 10) and increased concentration of SiO₂ (Figure S6 in Supporting Information S1), reflected in a modal increase of garnet (grossular). This suggests garnet grows by infiltration of aluminous and siliceous fluids sourced in the dehydrating oceanic lithosphere (Figure 10d) according to the simplified reaction:

$$3 \operatorname{Al}_2 O_3(aq) + 9 \operatorname{SiO}_2(s/aq) + 9 \operatorname{CaCO}_3(s) = 3 \operatorname{Ca}_3 \operatorname{Al}_2(\operatorname{SiO}_4)_3(s) + 9 \operatorname{CO}_2(aq)$$
(6)

where $Al_2O_3(aq)$ and $SiO_2(s/aq)$ denotes the presence of reactive Al and Si in the infiltrating fluid. The concentration of Al in those high-*P* fluids is greatly enhanced by forming (alkali)-Al/Si clusters (Anderson & Burnham, 1983; Manning, 2007; Manning et al., 2010), here represented by the species $AlO_2[SiO_2]^-$, cf. Table S4 in Supporting Information S1). This illustrates that the fate of carbonate, which is here directly linked to the potentials $\mu(CaO)$ and $\mu(CO_2)$ (e.g., Equation 4), in fact hinges on multicomponent interactions in the bulk system. This is because CaO associates with SiO_2 and a range of other components such as MgO, FeO and Al_2O_3 in minerals such as epidote, lawsonite, clinopyroxene, and garnet, as well as in carbonate solid solutions such as dolomite or ankerite (Lefeuvre et al., 2020) depending on *P*, *T*, and bulk composition. And since all those rock components are affected, to some degree, by metasomatic mass transfer (Figures 3 and 9) they all indirectly contribute to regulate $\mu(CO_2)$ (i.e., C solubility), and therefore the fate of carbon in those lithologies. We have not yet examined whether Mg and/or Fe cation substitution in carbonate plays an important role in the fluid-driven remobilization of CaO, as reported by Lefeuvre et al., 2020 in the western Alps.

An important implication of metasomatic growth of garnet in the subducting limestones of the central-eastern Pacific slabs, along with their progressive depletion in alkali and volatile (Figure 10), is that it would make those rocks much denser, and less buoyant, than previously thought (Behn et al., 2011; Ducea et al., 2022; Gerya & Meilick, 2011). In addition, assuming fluids tend to incorporate the lighter isotopes, our findings may also help understand the light isotopic signature of Ca (Antonelli et al., 2021) or Mg (Shen et al., 2018) in the metasomatized mantle wedge. Indeed, when an element is partially dissolved from the slab and released to the mantle wedge, which we show is most of the time the case (e.g., Si, Al, Ca, and Mg, Table 2, Figure 8), the isotopic signature of subduction fluid departs from that of the bulk slab, and becomes a complex function of fluid/rock ratio, mineral assemblage, temperature, and kinetics of dissolution reactions.

4.2. Sensitivity and Thermomechanical Limitations of Model

Our model captures key aspects of the thermodynamics of fluid-rock interactions in subduction zones, but a vertical transport of fluid does not adequately reflect the natural environment in multiple ways. Future works may explore the effect of pressure gradients (Wang, 2019; Wilson et al., 2014), more complex advection-dispersion designs (Ague, 2007), and channelized fluid flow which may apply to zones of high permeability contrasts

(Breeding et al., 2004), lithological interfaces (Angiboust et al., 2014), shear zone (Mancktelow, 2006), (micro) fractures, and connected porosity (wave; Plümper et al., 2017; Tian et al., 2019). While lateral fluid flow may significantly modify our fore-arc budgets, they may also create situations where only a fraction of the fluid reacts with the rock it traverses (as opposed to 100% in our model) while moving along a preferential flow path. We have tested this semi-quantitatively in the case of Cascadia, Nicaragua, and Honshu (Figure S14 in Supporting Information S1). The results suggest that reducing the reactive volume by 50% tends to reduce the amount of carbon released from the slabs by ca. 20%–30% before subarc depth. As mentioned previously, Penniston-Dorland et al. (2015) argued that slabs may be hotter than predicted by thermomechanical models (by > 300° C). Our semi-quantitative tests (Table S7 and Figures S14, S21 and S22 in Supporting Information S1) show that increasing the temperature by +50 and +100°C increases the carbon release by ca. 10%–30%. In other words, temperature effects may counter those of a channelization of fluid flow.

Computational aspects are relevant as well. By definition, none of the existing variants of the backcalculation algorithm (Galvez et al., 2015; Connolly & Galvez, 2018) include electrolytes in the search for the Gibbs energy minimum of the rock/solvent system. Although it is highly precise in most rock-dominated situations (i.e., where rock/electrolyte ratio is very high), a backcalculation strategy necessarily loses precision when specific components reach vanishingly low concentrations in the rock while it is present in the low volume of infiltrating fluid (e.g., specific fluid-dominated condition only occurs here in the case of K₂O, H₂O and CO₂ for Nicaragua and Mexico sediments). Practically, this means our algorithm may sometimes fail in refining low amounts of phengite and carbonate that may form transiently in these situations. However, this is inconsequential here because (a) the mass involved are always negligible, and, (b) more importantly, mass balance is maintained at all times. That is, the absence of suitable solid hosts for a specific component results here in its conservative, and out-of-equilibrium, transfer from the basaltic layer to the top of the slab in soluble form (e.g., potassic fluid in K-free sediment, Figures S2 and S3 in Supporting Information S1), with minimal interaction in the sediment. The implication is that the composition of a natural rock may not faithfully record the composition of an infiltrating fluid if some fluid components do not have a thermodynamically stable host in the rock. Thermodynamically more rigorous alternatives to our approach exist and would include all solids, solutes, and solvents in finding the Gibbs energy minimum of the system. The softwares HCh/HighPGibbs (Shvarov, 2008; Zhong et al., 2020) or GEM-Selektor (Kulik et al., 2013) implement such comprehensive GEM approach, but their algorithmic advantage (in part designed to address low-P systems were fluid-rock ratio are typically high) comes at the cost of a more limited solid solution package, which currently restricts their applicability to high-P systems were multicomponent minerals and solid-solutions predominate. The steep computational cost of such strategies is deemed unjustified for the present purpose. Indeed, we believe the main obstacle to accurate prediction of high-P fluid compositions is not algorithmic anymore, but experimental.

The accuracy of our fluid composition and stoichiometry depends on the quality of our thermochemical databases and of the experiments used to build them (Methods, Table S4 in Supporting Information S1). Our predictions for the solubility of the assemblage albite-paragonite-quartz (Al, Si, and Na concentrations in fluid, Figure S16 in Supporting Information S1) are slightly underestimated compared to the experimental data from Manning et al. (2010), but the orders of magnitude are correct over the P-T range considered. Increasing difference with T is caused by the progressive polymerization of alkali-alumino-silicate clusters (Anderson & Burnham, 1983; Newton & Manning, 2008; Pokrovski, 1998). The recent experiments on aragonite solubility (Facq et al., 2014) and their integration within the thermodynamic database of Ca-electrolytes, ensures that the Ca solubilities in our system—where C speciation is controlled by Ca-C complexes (cf. Figure S2 in Supporting Information S1)—are also probably relatively accurate. However, there is no equivalent data for Mg-carbonates and for the thermodynamics of Mg-bearing electrolytes. This means that unlike Ca, Mg concentration and Mg fluxes out of slabs may be underestimated by about 2 orders of magnitude or more (Galvez et al., 2015), and that our predicted solid solutions may bias toward their Mg-endmembers. The solubility of Fe-bearing phases, with their well-known redox sensitivity, may be the least well constrained of all. In this respect, it is worth noting that the solubility of corundum (Al₂O₃) increases by about 2 orders of magnitude in the presence of quartz, at 700 $^{\circ}$ C and 10 kbar (Manning (2007)). By analogy with the Al^{3+} cation, we conjecture that Fe^{3+} may be readily dissolved and transported as Si or Al-Si-complexes in high P-T metasomatic environments. The experiments of Pokrovski et al. (2003) support this hypothesis. Similarly, as a divalent cation like Ca^{2+} and Mg^{2+} , Fe^{2+} should readily form complex with dissolved silica and carbonate, for example, but none of those species are currently included in our model. Fe concentration in modeled metamorphic fluids may therefore be currently underestimated by about 2



orders of magnitude at least, and our predicted solid solutions may bias toward Fe-rich compositions. This hints to a set of relevant experimental data that would enhance the predictive accuracy of thermopetrological models, and allow exploring more complex redox systems (e.g., Evans & Frost, 2020; Galvez et al., 2013; Galvez & Jaccard, 2021). More generally, we note that accurate element solubility predictions ensure accurate prediction of its mobility/compatibility relation in the (open) fluid/rock system. But accurate fluid stoichiometry (i.e., element ratio in the fluid) is often more important because it determines the accuracy of mineralogical compositions (relative ratio of elements within a given phase, regardless of the modal abundance of the phase).

As for the accuracy of global elemental budgets, the above limitations appear relatively minor in front of the most critical assumption of our simulation, which is mechanical. We have made the common assumption that slabs are mechanically continuous, but the geological record and theoretical studies suggest this is certainly not the case in general (Malavieille & Ritz, 1989; Platt, 1986) as there are many non-advective (fluid-mediated) ways to remove material from the slab and export it to the wedge. Possibilities include the buoyant rise of subducted slabs of crustal material relative to higher density mantle rocks (Ernst, 1975; Gerya & Meilick, 2011; Molnar & Gray, 1979; Platt, 1986), and underplating of material at the base of accretionary wedge or of the continental mantle (Angiboust et al., 2022; Chakraborty et al., 1995; Cowan & Silling, 1978; Platt, 1986). For example, the metasedimentary units exposed in the Mediterranean region (Apennines, western and central Alps etc.) are built on top of Triassic evaporitic successions. These weak evaporitic layers in the western Alps (Barré et al., 2020; Gabalda et al., 2009), and elsewhere (Molli et al., 2020; Ring & Layer, 2003), favor detachment and exhumation of the sedimentary cover, and preclude its subduction (Malavieille & Ritz, 1989; Menant et al., 2020; Platt, 1986). Evidence for active underplating in the 25-45 km depth range have also been reported in today's Cascadia, Northern Chile, Alaska, and other subduction zones of the circumpacific area, and reviewed recently (Angiboust et al., 2022). Within the context of this study, this means that an unquantified fraction of mass transport to the mantle wedge may occur mechanically/gravitationally, possibly promoted by hybridization at the slab/mantle interface (Bebout & Barton, 1993; Sorensen et al., 2010), and not by advective transport.

5. Conclusions

Conventional petrological modeling has assumed slabs as consisting of discrete, compositionally homogeneous layers distinguished by sharp compositional transitions and exchanging H_2O and CO_2 only. By contrast, our model leads to intriguing patterns where layer boundaries become diffuse once advection of mass is considered. This is made possible by considering both volatiles and nonvolatiles as mobile components in the fluid phase. We found that the mobility of volatile and non-volatile elements is interdependent due to the formation of multicomponent aqueous species, and to the importance of multicomponent mineral and solid-solutions at elevated pressure and temperature.

Our slab operates as a chromatographic column in which elements are redistributed according to their respective affinity, or not, with the local rocks. As a result, rock compositions deviate significantly by metasomatism. Al, Si metasomatism, for example, drives the decarbonation of subducting limestones of the Central-eastern Pacific, and the synchronous growth of metasomatic garnet and loss of alkali and volatiles from these systems should form denser and less buoyant assemblages than previously thought. Conversely, some slabs accumulate alkali, and maintain them across dry segments of their trajectory, suggesting they may be metastable with respect to a fluid phase until deep in the subduction.

The fluids produced in the altered oceanic lithosphere gradually loose their ultramafic/mafic character as they move up in the rock column, and they reach the top of the slab with distinct sedimentary signatures. This means slab sediments are overall efficient at buffering fluid compositions before they reach and metasomatize the mantle wedge, at least until ca. 150 km depth. Of course, the present modelling framework may be applied to other types of lithological interfaces of local or global geological relevance (e.g., sediment-ultramafic interfaces). However, the constraints that our study brings on the compositional evolution of slabs is key because the composition of subducted rocks influences their rheology and mechanical properties.

Our forearc carbon flux ranges from 3% to 17% of the trench input, in large part due to its pronounced sensitivity to the choice of forearc-subarc threshold depth. This suggests that the existing discrepancy between model predictions and field observations of carbon release at fore-arc depths may be caused by the complex hydrodynamic regime of the shallow slabs rather than by inaccurate thermodynamics. The subarc flux represents 51%–62% of



the input, or 26–32 Mt C/y, which supports the existing notion of a high carbonate stability in subduction zones today. But we found that local heterogeneities between subduction zones are the rule rather than the exception. Moreover, this result does not mean much carbon is mechanically subducted beyond subarc depth, let alone the transition zone. This is because the chemical budget of our slabs depends on their structure and mechanical behavior, that is, that they move as a single mechanically continuous unit. This common assumption of thermo-dynamic-petrological simulations does not fully represent the observational, analogical, or thermomechanical evidences suggesting that sedimentary layers (pelagic or continental) may be detached from slabs along weak and ductile evaporitic layers, and underplated.

From a broader perspective, this work poses the question as to what part of the slab contribution to the mantle wedge is caused by fluid-driven mass advection, as simulated here, or by gravitational transport, for example by mantle hybridization, buoyancy forces and/or underplating. These issues must await future theoretical and experimental efforts.

Data Availability Statement

All the data have been provided in the figures and tables in the main text or the supplementary materials, which are also available at Open Science Framework osf.io/y84d2/.

References

- Abart, R. (1995). Phase equilibrium and stable isotope constraints on the formation of metasomatic garnet-vesuvianite veins (SW Adamello, N Italy). *Contributions to Mineralogy and Petrology*, 122, 116–133. https://doi.org/10.1007/s004100050116
- Agard, P. (2021). Subduction of oceanic lithosphere in the Alps: selective and archetypal from (slow-spreading) oceans. *Earth-Science Reviews*, 214, 103517.
- Ague, J. J. (2003). Fluid infiltration and transport of major, minor, and trace elements during regional metamorphism of carbonate rocks, Wepawaug Schist, Connecticut, USA. American Journal of Science, 303, 753–816. https://doi.org/10.2475/ajs.303.9.753
- Ague, J. J. (2007). Models of permeability contrasts in subduction zone mélange: Implications for gradients in fluid fluxes, Syros and Tinos Islands, Greece. *Chemical Geology*, 239, 217–227. https://doi.org/10.1016/j.chemgeo.2006.08.012
- Ague, J. J. (2011). Extreme channelization of fluid and the problem of element mobility during Barrovian metamorphism. *American Mineralogist*, *96*, 333–352. https://doi.org/10.2138/am.2011.3582
- Ague, J. J., & Nicolescu, S. (2014). Carbon dioxide released from subduction zones by fluid-mediated reactions. *Nature Geoscience*, 7, 355–360. https://doi.org/10.1038/ngeo2143
- Aitchison, J. (1982). The statistical analysis of compositional data. Journal of the Royal Statistical Society, 44, 139–160. https://doi. org/10.1111/j.2517-6161.1982.tb01195.x
- Alt, J. C., & Honnorez, J. (1984). Alteration of the upper oceanic crust, DSDP site 417: Mineralogy and chemistry. Contributions to Mineralogy and Petrology, 87, 149–169. https://doi.org/10.1007/BF00376221
- Alt, J. C., & Teagle, D. A. H. (2003). Hydrothermal alteration of upper oceanic crust formed at a fast-spreading ridge: Mineral, chemical, and isotopic evidence from ODP Site 801. *Chemical Geology*, 201, 191–211. https://doi.org/10.1016/S0009-2541(03)00201-8
- Anderson, G. M., & Burnham, C. W. (1983). Feldspar solubility and the transport of aluminum under metamorphic conditions. *American Journal of Science*, 283, 283–297.
- Angiboust, S., Glodny, J., Cambeses, A., Raimondo, T., Monié, P., Popov, M., & Garcia-Casco, A. (2021). Drainage of subduction interface fluids into the forearc mantle evidenced by a pristine jadeitite network (Polar Urals). *Journal of Metamorphic Geology*, 39, 473–500. https:// doi.org/10.1111/jmg.12570
- Angiboust, S., Menant, A., Gerya, T., & Oncken, O. (2022). The rise and demise of deep accretionary wedges: A long-term field and numerical modeling perspective. *Geosphere*, 18, 69–103. https://doi.org/10.1130/ges02392.1
- Angiboust, S., Muñoz-Montecinos, J., Cambeses, A., Raimondo, T., Deldicque, D., & Garcia-Casco, A. (2021). Jolts in the Jade factory: A route for subduction fluids and their implications for mantle wedge seismicity. *Earth-Science Reviews*, 220, 103720. https://doi.org/10.1016/j. earscirev.2021.103720
- Angiboust, S., Pettke, T., De Hoog, J. C. M., Caron, B., & Oncken, O. (2014). Channelized fluid flow and eclogite-facies metasomatism along the subduction shear zone. *Journal of Petrology*, 55, 883–916. https://doi.org/10.1093/petrology/egu010
- Antonelli, M. A., Kendrick, J., Yakymchuk, C., Guitreau, M., Mittal, T., & Moynier, F. (2021). Calcium isotope evidence for early Archaean carbonates and subduction of oceanic crust. *Nature Communications*, *12*. https://doi.org/10.1038/s41467-021-22748-2
- Aranovich, L. Y., & Newton, R. C. (1999). Experimental determination of CO₂-H₂O activity-composition relations at 600–1000°C and 6–14 kbar by reversed decarbonation and dehydration reactions. *American Mineralogist*, 84, 1319–1332. https://doi.org/10.2138/am-1999-0908
- Barré, G., Strzerzynski, P., Michels, R., Guillot, S., Cartigny, P., Thomassot, E., et al. (2020). Tectono-metamorphic evolution of an evaporitic décollement as recorded by mineral and fluid geochemistry: The "Nappe des Gypses" (Western Alps) case study. *Lithos*, 358, 105419. https:// doi.org/10.1016/j.lithos.2020.105419
- Barry, P. H., de Moor, J. M., Giovannelli, D., Schrenk, M., Hummer, D. R., Lopez, T., et al. (2019). Forearc carbon sink reduces long-term volatile recycling into the mantle. *Nature*, 568, 487–492. https://doi.org/10.1038/s41586-019-1131-5
- Bebout, G. E. (2007). Metamorphic chemical geodynamics of subduction zones. *Earth and Planetary Science Letters*, 260, 373–393. https://doi.org/10.1016/j.epsl.2007.05.050
- Bebout, G. E., & Barton, M. D. (1993). Metasomatism during subduction: Products and possible paths in the Catalina Schist, California. *Chemical Geology*, 108, 61–92. https://doi.org/10.1016/0009-2541(93)90318-D
- Behn, M. D., Kelemen, P. B., Hirth, G., Hacker, B. R., & Massonne, H. J. (2011). Diapirs as the source of the sediment signature in arc lavas. *Nature Geoscience*, 4(9), 641–646.

Acknowledgments

We thank Ellen Syracuse for sharing the thermal data of subduction zone and Terry Plank for discussions about uncertainties in sediment compositions. X.Z. acknowledges the financial support from Swiss National Science Foundation (P2EZP2_172220) and Alexander von Humboldt postdoc fellowship to finish this project. M.G. acknowledges support from the BrancoWeiss/Society in Science fellowship. We thank Tian Meng and an anonymous reviewer for constructive and insightful reviews. We thank Mark Dekkers and Stephen Parman for their editorial handling. We are grateful to Olivier Bachmann, Manuele Faccenda, Samuel Angiboust, Timm John, Peter Ulmer, Peter Clift, Philippe Agard, as well as graduate students Mathilde Joray and Nicole Helbing for stimulating discussions at various stages of this project. Open access funding enabled and organized by Projekt DEAL.

- Brady, J. B. (1977). Metasomatic zones in metamorphic rocks. Geochimica et Cosmochimica Acta, 41, 113-125. https://doi. org/10.1016/0016-7037(77)90191-0
- Breeding, C. M., Ague, J. J., & Bröcker, M. (2004). Fluid-metasedimentary rock interactions in subduction-zone mélange: Implications for the chemical composition of arc magmas. *Geology*, 32, 1041–1044. https://doi.org/10.1130/G20877.1
- Bucher, K. (1999). Growth mechanisms of metasomatic reaction veins in dolomite marbles from the Bergell Alps. *Mineralogy and Petrology*, 63, 151–171. https://doi.org/10.1007/bf01164149
- Caciagli, N. C., & Manning, C. E. (2003). The solubility of calcite in water at 6–16 kbar and 500–800°C. Contributions to Mineralogy and Petrology, 146, 275–285. https://doi.org/10.1007/s00410-003-0501-y
- Caldeira, K. (1991). Continental-pelagic carbonate partitioning and the global carbonate-silicate cycle. *Geology*, *19*, 204–206. https://doi.org/10 .1130/0091-7613(1991)019<0204:cpcpat>2.3.co;2
- Canales, J. P., Carbotte, S. M., Nedimovic, M. R., & Carton, H. (2017). Dry Juan de Fuca slab revealed by quantification of water entering Cascadia subduction zone. *Nature Geoscience*, 10, 864–870. https://doi.org/10.1038/NGEO3050
- Chakraborty, S., Dingwell, D. B., & Rubie, D. C. (1995). Multicomponent diffusion in ternary silicate melts in the system K₂O-A1₂O₃-SiO₂: II. Mechanisms, systematics, and geological applications. *Geochimica et Cosmochimica Acta*, 59, 265–277. https://doi. org/10.1016/0016-7037(95)00284-7
- Collins, N. C., Bebout, G. E., Angiboust, S., Agard, P., Scambelluri, M., Crispini, L., & John, T. (2015). Subduction zone metamorphic pathway for deep carbon cycling: II. Evidence from HP/UHP metabasaltic rocks and ophicarbonates. *Chemical Geology*, 412, 132–150. https://doi. org/10.1016/j.chemgeo.2015.06.012
- Connolly, J. A. D. (1990). Multivariable phase diagrams: An algorithm based on generalized thermodynamics. American Journal of Science, 290, 666–718. https://doi.org/10.2475/ajs.290.6.666
- Connolly, J. A. D. (2005). Computation of phase equilibria by linear programming: A tool for geodynamic modeling and its application to subduction zone decarbonation. *Earth and Planetary Science Letters*, 236, 524–541. https://doi.org/10.1016/j.epsl.2005.04.033
- Connolly, J. A. D. (2009). The geodynamic equation of state: What and how. *Geochemistry, Geophysics, Geosystems, 10*, Q10014. https://doi.org/10.1029/2009GC002540
- Connolly, J. A. D., & Galvez, M. E. (2018). Electrolytic fluid speciation by Gibbs energy minimization and implications for subduction zone mass transfer. *Earth and Planetary Science Letters*, 501, 90–102. https://doi.org/10.1016/j.epsl.2018.08.024
- Cook-Kollars, J., Bebout, G. E., Collins, N. C., Angiboust, S., & Agard, P. (2014). Subduction zone metamorphic pathway for deep carbon cycling: I. Evidence from HP/UHP metasedimentary rocks, Italian Alps. *Chemical Geology*, 386, 31–48. https://doi.org/10.1016/j. chemgeo.2014.07.013
- Cowan, D. S., & Silling, R. M. (1978). A dynamic, scaled model of accretion at trenches and its implications for the tectonic evolution of subduction complexes. *Journal of Geophysical Research*, 83, 5389. https://doi.org/10.1029/jb083ib11p05389
- Daswani, M., Vance, S. D., Mayne, M. J., & Glein, C. R. (2021). A metamorphic origin for Europa's ocean. Geophysical Research Letters, 48, 1–14. https://doi.org/10.1029/2021GL094143
- Ducea, M. N., Currie, C. A., Balica, C., Lazar, I., Mallik, A., Petrescu, L., & Vlasceanu, M. (2022). Diapirism of carbonate platforms subducted into the upper mantle. *Geology*.
- Ernst, W. G. (1975). Systematics of large-scale tectonics and age progressions in Alpine and Circum-Pacific blueschist belts. *Tectonophysics*, 26, 229–246. https://doi.org/10.1016/0040-1951(75)90092-X
- Evans, K. A., & Frost, B. R. (2020). Deserpentinization in subduction zones as a source of oxidation in arcs: A reality check. *Journal of Petrology*, 62(1), 1–32.
- Facq, S., Daniel, I., Montagnac, G., Cardon, H., & Sverjensky, D. A. (2014). In situ Raman study and thermodynamic model of aqueous carbonate speciation in equilibrium with aragonite under subduction zone conditions. *Geochimica et Cosmochimica Acta*, 132, 375–390. https://doi. org/10.1016/j.gca.2014.01.030
- Facq, S., Daniel, I., Montagnac, G., Cardon, H., & Sverjensky, D. A. (2016). Carbon speciation in saline solutions in equilibrium with aragonite at high pressure. *Chemical Geology*, 431, 44–53.
- Fein, J. B., & Walther, J. V. (1987). Calcite solubility in supercritical CO₂H₂O fluids. Geochimica et Cosmochimica Acta, 51, 1665–1673. https:// doi.org/10.1016/0016-7037(87)90346-2
- Fernández, D. P., Goodwin, A. R. H., Lemmon, E. W., Levelt Sengers, J. M. H., & Williams, R. C. (1997). A formulation for the static permittivity of water and steam at temperatures from 238 K to 873 K at pressures up to 1200 MPa, including derivatives and Debye-Hückel coefficients. *Journal of Physical and Chemical Reference Data*, 26, 1125–1166. https://doi.org/10.1063/1.555997
- Fernández, D. P., Mulev, Y., Goodwin, A. R. H., & Sengers, J. M. H. L. (1995). A database for the static dielectric constant of water and steam. Journal of Physical and Chemical Reference Data, 24, 33–70. https://doi.org/10.1063/1.555977
- Ferry, J. M. (1979). A map of chemical potential differences within an outcrop. American Mineralogist, 64, 966–985.
- Ferry, J. M., & Dipple, G. M. (1991). Fluid flow, mineral reactions, and metasomatism. *Geology*, 19, 211–214. https://doi.org/10.1130/0091-76 13(1991)019<0211:FFMRAM>2.3.CO;2
- Ferry, J. M., Ushikubo, T., & Valley, A. W. (2011). Formation of forsterite by silicification of dolomite during contact metamorphism. *Journal of Petrology*, 52, 1619–1640. https://doi.org/10.1093/petrology/egr021
- Fischer, T. P. (2008). Fluxes of volatiles (H₂O, CO₂, N₂, Cl, F) from arc volcanoes. *Geochemical Journal*, 42, 21–38. https://doi.org/10.2343/ geochemj.42.21
- Fischer, T. P., Arellano, S., Carn, S., Aiuppa, A., Galle, B., Allard, P., et al. (2019). The emissions of CO₂ and other volatiles from the world's subaerial volcanoes. *Scientific Reports*, 9, 1–11. https://doi.org/10.1038/s41598-019-54682-1
- Frezzotti, M. L., & Ferrando, S. (2015). The chemical behavior of fluids released during deep subduction based on fluid inclusions. American Mineralogist, 100(2-3), 352–377. https://doi.org/10.2138/am-2015-4933
- Gabalda, S., Beyssac, O., Jolivet, L., Agard, P., & Chopin, C. (2009). Thermal structure of a fossil subduction wedge in the Western Alps. *Terra Nova*, 21, 28–34. https://doi.org/10.1111/j.1365-3121.2008.00849.x
- Galvez, M. E., Connolly, J. A. D., & Manning, C. E. (2016). Implications for metal and volatile cycles from the pH of subduction zone fluids. *Nature*, 539, 420–424. https://doi.org/10.1038/nature20103
- Galvez, M. E., & Jaccard, S. L. (2021). Redox capacity of rocks and sediments by high temperature chalcometric titration. *Chemical Geology*, 564, 120016. https://doi.org/10.1016/j.chemgeo.2020.120016
- Galvez, M. E., Manning, C. E., Connolly, J. A. D., & Rumble, D. (2015). The solubility of rocks in metamorphic fluids: A model for rock-dominated conditions to upper mantle pressure and temperature. *Earth and Planetary Science Letters*, 430, 486–498. https://doi.org/10.1016/j.epsl.2015.06.019



- Galvez, M. E., Martinez, I., Beyssac, O., Benzerara, K., Agrinier, P., & Assayag, N. (2013). Metasomatism and graphite formation at a lithological interface in Malaspina (Alpine Corsica, France). *Contributions to Mineralogy and Petrology*, 166, 1687–1708. https://doi.org/10.1007/ s00410-013-0949-3
- Galvez, M. E., & Pubellier, M. (2019). How do subduction zones regulate the carbon cycle? In *Deep carbon: Past to present* (pp. 276–312). Cambridge University Press. https://doi.org/10.1017/9781108677950.010
- Gerya, T. V., & Meilick, F. I. (2011). Geodynamic regimes of subduction under an active margin: Effects of rheological weakening by fluids and melts. Journal of Metamorphic Geology, 29, 7–31. https://doi.org/10.1111/j.1525-1314.2010.00904.x
- Gorce, J. S., Caddick, M. J., & Bodnar, R. J. (2019). Thermodynamic constraints on carbonate stability and carbon volatility during subduction. *Earth and Planetary Science Letters*, 519, 213–222.
- Gorman, P. J., Kerrick, D. M., & Connolly, J. A. D. (2006). Modeling open system metamorphic decarbonation of subducting slabs. *Geochemistry, Geophysics, Geosystems*, 7, Q04007. https://doi.org/10.1029/2005GC001125
- Gresens, R. L. (1967). Composition-volume relationships of metasomatism. *Chemical Geology*, 2, 47–65. https://doi.org/10.1016/0009-2541(67)90004-6
- Hacker, B. R. (2008). H₂O subduction beyond arcs. *Geochemistry, Geophysics, Geosystems*, 9, Q03001. https://doi.org/10.1029/2007GC001707
 Harker, R. I., & Tuttle, O. F. (1956). Experimental data on the P (sub co 2)-T curve for the reaction; calcite-quartzwollastonite-carbon dioxide. *American Journal of Science*, 254, 239–256. https://doi.org/10.2475/ajs.254.4.239
- Harlow, G. E., & Sorensen, S. S. (2005). Jade (nephrite and jadeitite) and serpentinite: Metasomatic connections. *International Geology Review*, 47(2), 113–146.
- Hermann, J., Zheng, Y.-F., & Rubatto, D. (2013). Deep fluids in subducted continental crust. *Elements*, 9(4), 281–287. https://doi.org/10.2113/ gselements.9.4.281
- Hetényi, G., Cattin, R., Brunet, F., Bollinger, L., Vergne, J., Nábělek, J. L., & Diament, M. (2007). Density distribution of the India plate beneath the Tibetan plateau: Geophysical and petrological constraints on the kinetics of lower-crustal eclogitization. *Earth and Planetary Science Letters*, 264, 226–244. https://doi.org/10.1016/j.epsl.2007.09.036
- Hoareau, G., Bomou, B., Van Hinsbergen, D. J. J., Carry, N., Marquer, D., Donnadieu, Y., et al. (2015). Did high Neo-Tethys subduction rates contribute to early Cenozoic warming? *Climate of the Past*, 11, 1751–1767. https://doi.org/10.5194/cp-11-1751-2015
- Jarrard, R. D. (2003). Subduction fluxes of water, carbon dioxide, chlorine, and potassium. *Geochemistry, Geophysics, Geosystems*, 4, 8905. https://doi.org/10.1029/2002GC000392
- Joesten, R. (1977). Evolution of mineral assemblage zoning in diffusion metasomatism. *Geochimica et Cosmochimica Acta*, 41, 649–670. https:// doi.org/10.1016/0016-7037(77)90303-9
- John, T., Gussone, N., Podladchikov, Y. Y., Bebout, G. E., Dohmen, R., Halama, R., et al. (2012). Volcanic arcs fed by rapid pulsed fluid flow through subducting slabs. *Nature Geoscience*, 5, 489–492. https://doi.org/10.1038/ngeo1482
- Johnston, F. K. B., Turchyn, A. V., & Edmonds, M. (2011). Decarbonation efficiency in subduction zones: Implications for warm Cretaceous climates. *Earth and Planetary Science Letters*, 303, 143–152. https://doi.org/10.1016/j.epsl.2010.12.049
- Kawano, S., Katayama, I., & Okazaki, K. (2011). Permeability anisotropy of serpentinite and fluid pathways in a subduction zone. *Geology*, 39, 939–942. https://doi.org/10.1130/G32173.1

Kerrick, D. M. (1974). Review of metamorphic mixed-volatile (H₂O-CO₂) equilibrium. American Mineralogist, 59, 729–762.

- Kerrick, D. M., & Connolly, J. A. D. (1998). Subduction of ophicarbonates and recycling of CO₂ and H₂O. *Geology*, 26, 375–378. https://doi. org/10.1130/0091-7613(1998)026<0375:SOOARO>2.3.CO;2
- Kerrick, D. M., & Connolly, J. A. D. (2001a). Metamorphic devolatilization of subducted oceanic metabasalts: Implications for seismicity, arc magmatism and volatile recycling. *Earth and Planetary Science Letters*, 189, 19–29. https://doi.org/10.1016/S0012-821X(01)00347-8
- Kerrick, D. M., & Connolly, J. A. D. (2001b). Metamorphic devolatilization of subducted marine sediments and the transport of volatiles into the Earth's mantle. *Nature*, 411, 293–296. https://doi.org/10.1038/35077056
- Konrad-Schmolke, M., Halama, R., & Manea, V. C. (2016). Slab mantle dehydrates beneath Kamchatka—Yet recycles water into the deep mantle. Geochemistry, Geophysics, Geosystems, 17, 2987–3007. https://doi.org/10.1002/2016GC006335
- Kulik, D. A., Wagner, T., Dmytrieva, S. V., Kosakowski, G., Hingerl, F. F., Chudnenko, K. V., & Berner, U. R. (2013). GEM-Selektor geochemical modeling package: Revised algorithm and GEMS3K numerical kernel for coupled simulation codes. *Computers & Geosciences*, 17, 1–24. https://doi.org/10.1007/s10596-012-9310-6
- Lee, C. T. A., Shen, B., Slotnick, B. S., Liao, K., Dickens, G. R., Yokoyama, Y., et al. (2013). Continental arc-island arc fluctuations, growth of crustal carbonates, and long-term climate change. *Geosphere*, 9, 21–36. https://doi.org/10.1130/GES00822.1
- Lefeuvre, B., Agard, P., Verlaguet, A., Dubacq, B., & Plunder, A. (2020). Massive formation of lawsonite in subducted sediments from the Schistes Lustrés (W. Alps): Implications for mass transfer and decarbonation in cold subduction zones. *Lithos*, 370, 105629.
- Malavieille, J., & Ritz, J. F. (1989). Mylonitic deformation of evaporites in décollements: Examples from the southern Alps, France. Journal of Structural Geology, 11, 583–590. https://doi.org/10.1016/0191-8141(89)90089-8
- Malvoisin, B., Chopin, C., Brunet, F., & Galvez, M. E. (2012). Low-temperature wollastonite formed by carbonate reduction: A marker of serpentinite redox conditions. *Journal of Petrology*, 53(1), 159–176.
- Mancktelow, N. S. (2006). How ductile are ductile shear zones? Geology, 34, 345–348. https://doi.org/10.1130/G22260.1
- Manning, C. E. (1998). Fluid composition at the blueschist-eclogite transition in the model system Na₂O-MgO-Al₂O₃-SiO₂-H₂O-HCl. Schweizerische Mineralogische und Petrographische Mitteilungen, 78, 225–242. https://doi.org/10.5169/seals-59285
- Manning, C. E. (2004). The chemistry of subduction-zone fluids. Earth and Planetary Science Letters, 223, 1–16. https://doi.org/10.1016/j.epsl.2004.04.030
- Manning, C. E. (2007). Solubility of corundum + kyanite in H₂O at 700°C and 10 kbar: Evidence for Al-Si complexing at high pressure and temperature. *Geofluids*, 7, 258–269. https://doi.org/10.1111/j.1468-8123.2007.00179.x
- Mason, E., Edmonds, M., & Turchyn, A. V. (2017). Remobilization of crustal carbon may dominate volcanic arc emissions. Science, 357, 290–294. https://doi.org/10.1126/science.aan5049
- McGary, R. S., Evans, R. L., Wannamaker, P. E., Elsenbeck, J., & Rondenay, S. (2014). Pathway from subducting slab to surface for melt and fluids beneath Mount Rainier. *Nature*, *511*, 338–340. https://doi.org/10.1038/nature13493
- Menant, A., Angiboust, S., Gerya, T., Lacassin, R., Simoes, M., & Grandin, R. (2020). Transient stripping of subducting slabs controls periodic forearc uplift. *Nature Communications*, 11, 1–10. https://doi.org/10.1038/s41467-020-15580-7
- Menzel, M. D., Garrido, C. J., & López Sánchez-Vizcaíno, V. (2020). Fluid-mediated carbon release from serpentinite-hosted carbonates during dehydration of antigorite-serpentinite in subduction zones. *Earth and Planetary Science Letters*, 531, 115964. https://doi.org/10.1016/j.epsl.2019.115964

- Menzel, M. D., Garrido, C. J., López Sánchez-Vizcaíno, V., Hidas, K., & Marchesi, C. (2019). Subduction metamorphism of serpentinite-hosted carbonates beyond antigorite-serpentinite dehydration (Nevado-Filábride Complex, Spain). Journal of Metamorphic Geology, 37, 681–715. https://doi.org/10.1111/jmg.12481
- Miller, D. P., Marschall, H. R., & Schumacher, J. C. (2009). Metasomatic formation and petrology of blueschist-facies hybrid rocks from Syros (Greece): Implications for reactions at the slab-mantle interface. *Lithos*, 107, 53–67. https://doi.org/10.1016/j.lithos.2008.07.015
- Molli, G., Brogi, A., Caggianelli, A., Capezzuoli, E., Liotta, D., Spina, A., & Zibra, I. (2020). Late Palaeozoic tectonics in central Mediterranean: A reappraisal. Swiss Journal of Geosciences, 113, 1–32. https://doi.org/10.1186/s00015-020-00375-1
- Molnar, P., & Gray, D. (1979). Subduction of continental lithosphere. *Geology*, 7, 58–62. https://doi.org/10.1130/0091-7613(1979)7<58:soclsc>2.0.co;2
- Mottl, M. J., Wheat, C. G., Fryer, P., Gharib, J., & Martin, J. B. (2004). Chemistry of springs across the Mariana forearc shows progressive devolatilization of the subducting plate. *Geochimica et Cosmochimica Acta*, 68, 4915–4933. https://doi.org/10.1016/j.gca.2004.05.037
- Newton, R. C., & Manning, C. E. (2008). Thermodynamics of SiO₂-H2O fluid near the upper critical end point from quartz solubility measurements at 10 kbar. *Earth and Planetary Science Letters*, 274, 241–249. https://doi.org/10.1016/j.epsl.2008.07.028
- Pan, D., Spanu, L., Harrison, B., Sverjensky, D. A., & Galli, G. (2013). Dielectric properties of water under extreme conditions and transport of carbonates in the deep Earth. *Proceedings of the National Academy of Sciences of the United States of America*, 110, 6646–6650. https://doi. org/10.1073/pnas.1221581110
- Parai, R., & Mukhopadhyay, S. (2012). How large is the subducted water flux? New constraints on mantle regassing rates. *Earth and Planetary Science Letters*, 317–318, 396–406. https://doi.org/10.1016/j.epsl.2011.11.024
- Peacock, S. M. (1990). Processes subduction zones. Science, 329, 329–337. https://doi.org/10.1126/science.248.4953.329
- Penniston-Dorland, S. C., Walker, R. J., Pitcher, L., & Sorensen, S. S. (2012). Mantle-crust interactions in a paleosubduction zone: Evidence from highly siderophile element systematics of eclogite and related rocks. *Earth and Planetary Science Letters*. 319–320, 295–306. https:// doi.org/10.1016/j.epsl.2011.11.042
- Philippot, P., & Selverstone, J. (1991). Trace-element-rich brines in eclogitic veins: Implications for fluid composition and transport during subduction. Contributions to Mineralogy and Petrology, 106, 417–430. https://doi.org/10.1007/BF00321985
- Piccoli, F., Hermann, J., Pettke, T., Connolly, J. A. D., Kempf, E. D., & Vieira Duarte, J. F. (2019). Subducting serpentinites release reduced, not oxidized, aqueous fluids. *Scientific Reports*, 9, 1–7. https://doi.org/10.1038/s41598-019-55944-8
- Piccoli, F., Vitale Brovarone, A., Beyssac, O., Martinez, I., Ague, J. J., & Chaduteau, C. (2016). Carbonation by fluid-rock interactions at high-pressure conditions: Implications for carbon cycling in subduction zones. *Earth and Planetary Science Letters*, 445, 146–159. https://doi. org/10.1016/j.epsl.2016.03.045
- Plank, T., & Langmuir, C. H. (1998). The chemical composition of subducting sediment and its consequences for the crust and mantle. *Chemical Geology*, 145, 325–394. https://doi.org/10.1016/S0009-2541(97)00150-2

Plank, T., & Manning, C. E. (2019). Subducting carbon. Nature, 574, 343–352. https://doi.org/10.1038/s41586-019-1643-z

- Platt, J. P. (1986). Dynamics of orogenic wedges and the uplift of high-pressure metamorphic rocks. The Geological Society of America Bulletin, 97, 1037–1053. https://doi.org/10.1130/0016-7606(1986)97<1037:DOOWAT>2.0.CO;2
- Plümper, O., John, T., Podladchikov, Y. Y., Vrijmoed, J. C., & Scambelluri, M. (2017). Fluid escape from subduction zones controlled by channel-forming reactive porosity. *Nature Geoscience*, 10, 150–156. https://doi.org/10.1038/ngeo2865
- Pokrovski, G. S. (1998). Structure and stability of aluminum-silica complexes in neutral to basic solutions. Experimental study and molecular orbital calculations. *Mineralogical Magazine*, 62(2), 1194–1195.
- Pokrovski, G. S., Schott, J., Farges, F., & Hazemann, J. L. (2003). Iron (III)-silica interactions in aqueous solution: Insights from X-ray absorption fine structure spectroscopy. *Geochimica et Cosmochimica Acta*, 67(19), 3559–3573.
- Ring, U., & Layer, P. W. (2003). High-pressure metamorphism in the Aegean, eastern Mediterranean: Underplating and exhumation from the late Cretaceous until the Miocene to recent above the retreating Hellenic subduction zone. *Tectonics*, 22, 1022. https://doi. org/10.1029/2001TC001350
- Rumble, D. (1982). The role of perfectly mobile components in metamorphism. Annual Review of Earth and Planetary Sciences, 10, 221–233. https://doi.org/10.1146/annurev.ea.10.050182.001253
- Rustioni, G., Audetat, A., & Keppler, H. (2021). The composition of subduction zone fluids and the origin of the trace element enrichment in arc magmas. *Contributions to mineralogy and petrology*, 176(7), 1–19. https://doi.org/10.1007/s00410-021-01810-8
- Scambelluri, M., Cannaò, E., Guerini, S., Bebout, G. E., Epstein, G., Rotondo, F., & Tartarotti, P. (2022). Carbon Mobility and Exchange in a Plate-Interface Subduction Mélange: A Case Study of Meta-Ophiolitic Rocks in Champorcher Valley, Italian Alps. Available at SSRN: https:// ssrn.com/abstract=4044363 or https://doi.org/10.2139/ssrn.4044363
- Schmidt, M. W. (2015). Melting of pelitic sediments at subarc depths: 2. Melt chemistry, viscosities and a parameterization of melt composition. *Chemical Geology*, 404, 168–182. https://doi.org/10.1016/j.chemgeo.2015.02.013
- Schmidt, M. W., & Poli, S. (1998). Experimentally based water budgets for dehydrating slabs and consequences for arc magma generation. Earth and Planetary Science Letters, 163, 361–379. https://doi.org/10.1016/s0012-821x(98)00142-3
- Schmidt, M. W., & Poli, S. (2003). Generation of mobile components during subduction of oceanic crust. In *Treatise on geochemistry* (pp. 567–591). Elsevier. https://doi.org/10.1016/B0-08-043751-6/03034-6
- Schmidt, M. W., Vielzeuf, D., & Auzanneau, E. (2004). Melting and dissolution of subducting crust at high pressures: The key role of white mica. Earth and Planetary Science Letters, 228, 65–84. https://doi.org/10.1016/j.epsl.2004.09.020
- Schneider, M. E., & Eggler, D. H. (1986). Fluids in equilibrium with peridotite minerals: Implications for mantle metasomatism. Geochimica et Cosmochimica Acta, 50, 711–724. https://doi.org/10.1016/0016-7037(86)90347-9
- Shen, J., Li, S. G., Wang, S. J., Teng, F. Z., Li, Q. L., & Liu, Y. S. (2018). Subducted Mg-rich carbonates into the deep mantle wedge. Earth and Planetary Science Letters, 503, 118–130. https://doi.org/10.1016/j.epsl.2018.09.011
- Shvarov, Y. V. (2008). HCh: New potentialities for the thermodynamic simulation of geochemical systems offered by windows. *Geochemistry International*, 46, 834–839. https://doi.org/10.1134/S0016702908080089
- Sorensen, S. S., Sisson, V. B., Harlow, G. E., & Avé Lallemant, H. G. (2010). Element residence and transport during subduction-zone metasomatism: Evidence from a jadeitite-serpentinite contact, Guatemala. *International Geology Review*, 52, 899–940. https://doi. org/10.1080/00206810903211963
- Spandler, C., Mavrogenes, J., & Hermann, J. (2007). Experimental constraints on element mobility from subducted sediments using high-P synthetic fluid/melt inclusions. *Chemical Geology*, 239, 228–249. https://doi.org/10.1016/j.chemgeo.2006.10.005
- Staudigel, H., Hart, S. R., Schmincke, H. U., & Smith, B. M. (1989). Cretaceous ocean crust at DSDP Sites 417 and 418: Carbon uptake from weathering versus loss by magmatic outgassing. *Geochimica et Cosmochimica Acta*, 53, 3091–3094. https://doi.org/10.1016/0016-7037(89)90189-0



- Stolper, E., & Newman, S. (1994). The role of water in the petrogenesis of Mariana trough magmas. Earth and Planetary Science Letters, 121, 293–325. https://doi.org/10.1016/0012-821X(94)90074-4
- Sverjensky, D. A., Stagno, V., & Huang, F. (2014). Important role for organic carbon in subduction-zone fluids in the deep carbon cycle. Nature Geoscience, 7, 909–913. https://doi.org/10.1038/ngeo2291
- Tackley, P. (2000). Self-consistent generation of tectonic plates in time- dependent, three-dimensional mantle convection simulations. Geochemistry, Geophysics, Geosystems, 1(8), 1021. https://doi.org/10.1029/2000gc000036
- Tatsumi, Y., Hamilton, D. L., & Nesbitt, R. W. (1986). Chemical characteristics of fluid phase released from a subducted lithosphere and origin of arc magmas: Evidence from high-pressure experiments and natural rocks. *Journal of Volcanology and Geothermal Research*, 29, 293–309. https://doi.org/10.1016/0377-0273(86)90049-1
- Tian, M., Katz, R. F., Rees Jones, D. W., & May, D. A. (2019). Devolatilization of subducting slabs, part II: Volatile fluxes and storage. Geochemistry, Geophysics, Geosystems, 20, 6199–6222. https://doi.org/10.1029/2019GC008489
- Van Keken, P. E., Hacker, B. R., Syracuse, E. M., & Abers, G. A. (2011). Subduction factory: 4. Depth-dependent flux of H₂O from subducting slabs worldwide. *Journal of Geophysical Research*, 116, B01401. https://doi.org/10.1029/2010JB007922
- Vitale Brovarone, A., & Beyssac, O. (2014). Lawsonite metasomatism: A new route for water to the deep Earth. Earth and Planetary Science Letters, 393, 275–284. https://doi.org/10.1016/j.epsl.2014.03.001

Volk, T. (1989). Sensitivity of climate and atmospheric CO₂ to deep-ocean and shallow-ocean carbonate burial. *Nature*, 337, 637–640. https://doi.org/10.1038/337637a0

- Wang, H., Huismans, R. S., & Rondenay, S. (2019). Water migration in the subduction mantle wedge: A two-phase flow approach. Journal of Geophysical Research: Solid Earth, 124, 9208–9225. https://doi.org/10.1029/2018JB017097
- Wilson, C. R., Spiegelman, M., Keken, P. E. V., & Hacker, B. R. (2014). Fluid flow in subduction zones : The role of solid rheology and compaction pressure. *Earth and Planetary Science Letters*, 401, 261–274. https://doi.org/10.1016/j.epsl.2014.05.052
- Wolery, T. J. (1992). *EQ3/6, a software package for geochemical modeling of aqueous systems: Package overview and installation guide* (version 7.0) (No. UCRL-MA--110662-PT. 1). Lawrence Livermore National Lab.
- Zhong, R., Li, Y., Etschmann, B., Brugger, J., Yu, C., & Cui, H. (2020). HighPGibbs, a practical tool for fluid-rock thermodynamic simulation in deep Earth and its application on calculating nitrogen speciation in subduction zone fluids. *Geochemistry, Geophysics, Geosystems*, 21(5), e2020GC008973.

References From the Supporting Information

Anderson, G., & Crerar, D. A. (1993). Thermodynamics in geochemistry: The equilibrium model. Oxford University Press.

- Benson, S. W. (1976). Thermochemical kinetics. John Wiley and Sons.
- Cai, C., Wiens, D. A., Shen, W., & Eimer, M. (2018). Water input into the Mariana subduction zone estimated from ocean-bottom seismic data. *Nature*, 563, 389–392. https://doi.org/10.1038/s41586-018-0655-4
- Clift, P. D. (2017). A revised budget for Cenozoic sedimentary carbon subduction. Reviews of Geophysics, 55, 97-125. https://doi. org/10.1002/2016rg000531
- Coggon, R., & Holland, T. J. B. (2002). Mixing properties of phengitic micas and revised garnet-phengite thermobarometers. *Journal of Meta-morphic Geology*, 20, 683–696. https://doi.org/10.1046/j.1525-1314.2002.00395.x
- Connolly, J. A. D., & Trommsdorff, V. (1991). Petrogenetic grids for metacarbonate rocks: Pressure-temperature phase-diagram projection for mixed-volatile systems. *Contributions to Mineralogy and Petrology*, 108, 93–105. https://doi.org/10.1007/BF00307329
- Dale, J., Powell, R., White, R. W., Elmer, F. L., & Holland, T. J. B. (2005). A thermodynamic model for Ca-Na clinoamphiboles in Na₂O-CaO-FeO-MgO-Al₂O₃-SiO₂-H2O-O for petrological calculations. *Journal of Metamorphic Geology*, 23, 771–791. https://doi. org/10.1111/j.1525-1314.2005.00609.x
- Dasgupta, R., & Hirschmann, M. M. (2010). The deep carbon cycle and melting in Earth's interior. *Earth and Planetary Science Letters*, 298, 1–13. https://doi.org/10.1016/j.epsl.2010.06.039
- Fernandez, D., Goodwin, A., Lemmon, E. W., Sengers, J. L., & Williams, R. (1997). A formulation for the static permittivity of water and steam at temperatures from 238 K to 873 K at pressures up to 1200 MPa, including derivatives and Debye-Hückel coefficients. *Journal of Physical* and Chemical Reference Data, 26(4), 1125–1166.
- Galvez, M. E., Manning, C. E., Connolly, J. A. D., & Rumble, D. (2015). The solubility of rocks in metamorphic fluids: A model for rock-dominated conditions to upper mantle pressure and temperature. *Earth and Planetary Science Letters*, 430, 486–498. https://doi.org/10.1016/j. epsl.2015.06.019
- Hart, S. R., & Zindler, A. (1986). In search of a bulk-Earth composition. *Chemical Geology*, 57, 247–267. https://doi. org/10.1016/0009-2541(86)90053-7
- Helgeson, H. C., Delany, J. M., Nesbitt, H. W., & Bird, D. K. (1978). Summary and critique of the thermodynamic properties of rock-forming minerals. American Journal of Science, 278, 1–229.
- Helgeson, H. C., Kirkham, D. H., & Flowers, G. C. (1981). Theoretical prediction of the thermodynamic behavior of aqueous electrolytes at high pressures and temperatures: IV. Calculation of activity coefficients, osmotic coefficients, and apparent molal and standard and relative partial molal properties to 600°C. American Journal of Science, 281, 1249–1516. https://doi.org/10.2475/ajs.281.10.1249
- Hermann, J., Zheng, Y.-F., & Rubatto, D. (2013). Deep fluids in subducted continental crust. *Elements*, 9(4), 281–287. https://doi.org/10.2113/ gselements.9.4.281
- Holland, T., & Powell, R. (2003). Activity-compositions relations for phases in petrological calculations: An asymetric multicomponent formulation. Contributions to Mineralogy and Petrology, 145, 492–501. https://doi.org/10.1007/s00410-003-0464-z
- Holland, T., & Powell, R. (1996). Thermodynamics of order-disorder in minerals: I. Symmetric formalism applied to minerals of fixed composition. American Mineralogist, 81, 1413–1424. https://doi.org/10.2138/am-1996-11-1214
- Holland, T. J. B., & Powell, R. (1998). An internally consistent thermodynamic data set for phases of petrological interest. Journal of Metamorphic Geology, 16, 309–343.
- Johnson, J. W., Oelkers, E. H., & Helgeson, H. C. (1992). SUPCRT92: A software package for calculating the standard molal thermodynamic properties of minerals, gases, aqueous species, and reactions from 1 to 5000 bar and 0 to 1000°C. Computers & Geosciences.
- Kelemen, P. B., Manning, C. E. (2015). Reevaluating carbon fluxes in subduction zones, what goes down, mostly comes up. Proceedings of the National Academy of Sciences of the United States of America, 112, E3997–E4006. https://doi.org/10.1073/pnas.1507889112



- Manning, C. E., Antignano, A., & Lin, H. A. (2010). Premelting polymerization of crustal and mantle fluids, as indicated by the solubility of albite+paragonite+quartz in H₂O at 1 GPa and 350–620°C. *Earth and Planetary Science Letters*, 292, 325–336. https://doi.org/10.1016/j. epsl.2010.01.044
- Mountain, R. D., & Harvey, A. H. (2015). Molecular dynamics evaluation of dielectric constant mixing rules for H₂O-CO₂ at geologic conditions. *Journal of Solution Chemistry*, 44(11), 2179–2193. https://doi.org/10.1007/s10953-015-0401-6
- Padrón-Navarta, J. A., Sánchez-Vizcaíno, V. L., Hermann, J., Connolly, J. A. D., Garrido, C. J., Gómez-Pugnaire, M. T., & Marchesi, C. (2013). Tschermak's substitution in antigorite and consequences for phase relations and water liberation in high-grade serpentinites. *Lithos*, 178, 186–196. https://doi.org/10.1016/j.lithos.2013.02.001
- Pan, D., Spanu, L., Harrison, B., Sverjensky, D. A., & Galli, G. (2013). Dielectric properties of water under extreme conditions and transport of carbonates in the deep Earth. Proceedings of the National Academy of Sciences of the United States of America, 110(17), 6646–6650.
- Pearce, J. A. (1976). Statistical analysis of major element patterns in basalts. Journal of Petrology, 17, 15-43. https://doi.org/10.1093/ petrology/17.1.15
- Penniston-Dorland, S. C., Kohn, M. J., & Manning, C. E. (2015). The global range of subduction zone thermal structures from exhumed blueschists and eclogites : Rocks are hotter than models. *Earth and Planetary Science Letters*, 428, 243–254. https://doi.org/10.1016/j.epsl.2015.07.031 Plank, T. (2013). The chemical composition of subducting sediments. *Treatise on geochemistry* (2nd ed.). Elsevier Ltd.
- Rüpke, L. H., Morgan, J. P., Hort, M., & Connolly, J. A. D. (2004). Serpentine and the subduction zone water cycle. Earth and Planetary Science Letters, 223, 17–34. https://doi.org/10.1016/j.epsl.2004.04.018
- Shock, E. L., Oelkers, E. H., Johnson, J. W., Sverjensky, D. A., & Helgeson, H. C. (1992). Calculation of the thermodynamic properties of aqueous species at high pressures and temperatures. *Journal of the Chemical Society Faraday Transactions*, 88, 803–826. https://doi.org/10.1039/ ft9928800803
- Shock, E. L., & Helgeson, H. C. (1990). Calculation of the thermodynamic and transport properties of aqueous species at high pressures and temperatures: Standard partial molal properties of organic species. *Geochimica et Cosmochimica Acta*, 54, 915–945. https://doi. org/10.1016/0016-7037(90)90429-0
- Sverjensky, D. A., Harrison, B., & Azzolini, D. (2014). Water in the deep Earth: The dielectric constant and the solubilities of quartz and corundum to 60kb and 1200°C. *Geochimica et Cosmochimica Acta*, 129, 125–145. https://doi.org/10.1016/j.gca.2013.12.019
- Syracuse, E. M., van Keken, P. E., Abers, G. A., Suetsugu, D., Bina, C., Inoue, T., et al. (2010). The global range of subduction zone thermal models. *Physics of the Earth and Planetary Interiors*, 183, 73–90. https://doi.org/10.1016/j.pepi.2010.02.004
- Tajčmanová, L., Connolly, J. A. D., & Cesare, B. (2009). A thermodynamic model for titanium and ferric iron solution in biotite. Journal of Metamorphic Geology, 27, 153–165.
- Tanger, J. C., & Helgeson, H. C. (1988). Calculation of the thermodynamic and transport properties of aqueous species at high pressures and temperatures; revised equations of state for the standard partial molal properties of ions and electrolytes. *American Journal of Science*.
- Thompson, J. B., & Hovis, G. L. (1979). Entropy of mixing in sanidine. American Mineralogist, 64, 57-64. https://doi.org/10.1136/jme.5.2.57